

The challenge of adopting mitigation and adaptation measures for the impacts of sand and dust storms in Eastern Mediterranean Region: a critical review

Andreas Eleftheriou¹ · Petros Mouzourides¹ · George Biskos^{2,3} · Panayiotis Yiallouros⁴ · Prashant Kumar⁵ · Marina K.-A. Neophytou¹

Received: 21 April 2022 / Accepted: 29 May 2023 / Published online: 19 June 2023 © The Author(s) 2023, corrected publication 2023

Abstract

Sand and dust storms (SDS) are a major disruptor in both the source areas where they occur and at distant locations. This critical review aims to address the question of whether mitigation and adaptation measures have been or can be implemented and what is the optimal scale of their implementation to negate the impacts of SDS in Eastern Mediterranean Region (EMR)? Measures which differ in approach are also assessed by recording their successes, failures, and future challenges. We conclude that developing and implementing appropriate mitigation or adaptation measures for SDS at the local level is feasible but, at a wider scale, is a new challenge. This challenge is even more complex in areas like the EMR and the SDS sources affecting it, as it is a crossroad of air masses originating from three major SDS areas, which exhibit economic, political, and social diversity. This review also aims to identify successful mitigation strategies that have been used for similar environmental issues and to draw attention to the lack of adaptation measures in the region. This critical synthesis will serve as a guide for public stakeholders considering measures to mitigate or adapt to SDS based on their effectiveness and the area of implementation.

Keywords Desert particles · Erodible land surfaces · Future pressures · Socioeconomic and political dimensions · Transboundary air pollution

Andreas Eleftheriou eleftheriou.g.andreas@ucy.ac.cy

- ³ Faculty of Civil Engineering and Geosciences, Delft University of Technology, Delft, The Netherlands
- ⁴ Medical School, University of Cyprus, Nicosia, Cyprus
- ⁵ Global Centre for Clean Air Research (GCARE), Department of Civil and Environmental Engineering, Faculty of Engineering and Physical Sciences (FEPS), University of Surrey, Guildford GU2 7XH, UK

¹ Environmental Fluid Mechanics Laboratory, Department of Civil and Environmental Engineering, University of Cyprus, Nicosia, Cyprus

² Climate and Atmosphere Research Centre, The Cyprus Institute, Nicosia, Cyprus

1 Introduction

Air quality is of critical and rising importance, as widely acknowledged by the World Health Organization (WHO) and many other International and National Public Health Organizations, because 4.2 and 3.8 million premature deaths annually are associated to exposures to ambient and indoor air pollution, respectively (WHO 2016). Previous studies have attributed the increasing trend of both disability-adjusted life years and mortality in various areas of the world to the increasing emission rates of anthropogenic and natural air pollutants in relation to other factors such as population ageing (Babatola 2018; Cohen et al. 2017; Lelieveld et al. 2015).

Of all the major air pollutants that affect human health, particulate matter (PM) exhibits the most multi-scale and multi-variable behaviour (Betzer et al. 1988; van der Does et al. 2018), due to its multiple physicochemical characteristics and emission mechanisms. Emissions of sand and dust particles in the atmosphere usually emerge from hyper arid, arid, or semi-arid areas with barren surroundings and low or no vegetation, such as deserts, which are also characterised by the prevalence of strong winds and low or no moisture (Choobari et al. 2014; Ghose 2002; Middleton and Kang 2017; Thomas et. al. 2005; Webb and Pierre 2018). However, such sources are not located only in uninhabited areas, but also in places where many human activities occur. Agricultural areas in Niger that have been degraded due to droughts (Abdou 2013), open mining areas, and abandoned mines in Colombia and Australia (Huertas et al. 2014; Tordoff et al. 2000) are such distinctive examples. In particular, agricultural activities and overgrazing have caused a land change in vegetation and cover, which, under certain meteorological conditions of wind speed and moisture, lead to sand and dust storms (SDS) (Gill 1996; Liu et al. 2015). Interestingly, Mahowald et al. (2010) used ice, coral, and lake cores close to the main source areas (dust paleodata) to show that dust emissions doubled in the twentieth century because of human influence. However, as the same study indicates, other reports have recorded a -20 to +60% variation in the anthropogenic effect on desert dust emissions, demonstrating the uncertainty and importance of further research in this field.

The World Meteorological Organization (WMO) defines SDS as 'an ensemble of particles of dust or sand energetically lifted to higher elevations by a strong and turbulent wind' (WMO 2021). Many effects of moderate or high intensity SDS, namely the impact of particles on human health and various economic sectors, have been systematically studied (e.g. Tsiouri et al. 2015). However, associated mitigation and adaptation measures are rarely studied. For example, only Middleton and Kang's (2017) review and a large-scale report (UNEP et al. 2016) have gathered different impact mitigation and adaptation measures globally. The review paper and the report have not reported how directly affected countries in the Eastern Mediterranean Region (EMR) use or can use measures to adapt the phenomenon and how the economic, political, and social diversities in each area are important to be highlighted before implementing mitigation and adaptation measures.

The EMR and the sources affecting it are economically, politically, and culturally heterogeneous, and its geographic boundaries are not accurately defined in the literature. Therefore, challenges in implementing and evaluating mitigation and adaptation measures are even greater. In this review, the EMR refers to countries around the Eastern Mediterranean Sea, namely Cyprus, Greece, Israel, Turkey, Lebanon, Syria, and Egypt. This review examined mitigation measures in the source areas of the Middle East, Northern Africa, and the Sahel, where SDS originate and do not belong to the pre-defined countries of the EMR, whereas the adaptation measures are examined in the EMR countries as shown in Fig. 1. The term 'measure' refers to a single or a series of actions at a local or at a wider scale. The aim of this paper is to review the different mitigation and adaptation measures that are taken or can be taken



Fig. 1 The graphical representation of the assessment of mitigation and adaptation measures, as well as the presentation of the investigation areas

in the source areas of SDS or the EMR in order to negate the impacts of SDS locally or far away, while investigating the challenges of implementing the measures by the different social, economic, and political dimensions' future challenges.

The search strategy used to conduct this critical review included the Scopus, Web of Science, and Google Scholar databases. In addition, the websites of European and non-European environmental agencies were searched for case reports, editorials, scientific papers, and critical reviews published in English from 1984 to 2023. The keywords dust, impact mitigation, mitigation and adaptation measures, transboundary emissions, Eastern Mediterranean Region, Sahel, Middle East, and Northern Africa were used.

This paper is structured as follows: Section 2 provides the reasoning behind the investigation of SDS in the EMR; Section 3 briefly reviews the impacts of SDS; Sections 4 and 5 present the different mitigation and adaptation measures in the study area; Section 6 examines the socioeconomic and political aspects of implementing the measures; Section 7 discusses the future pressures that can affect the implementation of measures; and finally, Section 8 summarises the key conclusions of this review.

2 SDS in Eastern Mediterranean Region

DS have been reported since ancient times; the historian Herodotus reported that an SDS in 525 B.C. buried the army of the Cambyses (Herodotus III: 86–88) either from hearsay or the aftermath of the event. In the modern era, the methods of identification, monitoring, and analysis, as well as the understanding of the mechanisms that govern SDS, have improved. Despite the different characteristics of the sources, they share a common feature, which are the erodible surfaces. The wind erodibility of ground surfaces is controlled by various factors, such as the land cover (e.g. vegetation and rocks), characteristics of individual soil types (e.g. texture), or biological compositions, which can be of importance in sand and dust emissions (Katra et al. 2017; Soil and Water Conservation Society 2017; Zobeck and Van Pelt 2015). In addition, other characteristics, such as topographic depressions or locations near the downwind side of mountain ridges (Cuesta et al. 2009; Doyle and Durran 2002), can be of importance in the emission of sand and dust. Once the particles are lifted to higher elevations, they are transported over long distances, subject to the prevailing meteorological conditions. Finally, the suspended particles are brought to the earth's surface, depending on the prevailing atmospheric conditions, by dry and wet deposition processes (Knippertz and Stuut 2014). Figure 2 shows the lifetimes of SDS.

SDS in the EMR originate more frequently from Northern Africa and the Sahel (North Africa) during spring, and during autumn from the Middle East (Gherboudj et al. 2017; Varga et al. 2014; Kubilay et al. 2000). The sources are separated in 3 types: (a) hydrological, referring to playas, ephemeral, and dried lakes, (b) natural non-hydrological, referring to areas with low land use such as sand and dust dunes, and (c) anthropogenic, referring to areas with high land use such as agricultural fields (Ginoux et al. 2012). In North Africa, the Sahara Desert is spanning from the Atlantic Ocean to the Suez Canal and from the Mediterranean Sea to the Sahel, consisting of the largest sand and dust source in the world. More than half of the global dust emission budget, around 500–1400 Tg year⁻¹, is emitted from this desert (Ginoux et al. 2004; Miller et al.2004; Tanaka and Chiba 2006), while Gherboudj et al. (2017) reported that more than twenty (20) source areas of SDS have been identified across North Africa. The Bodele depression, the Nubian Desert in Egypt, and the Saharan Atlas slopes are only some of the high dust emission areas.

In addition, the Middle East is situated between the three (3) areas of the Global Dust Belt, North Africa, and Central Asia, both of which are key SDS sources. This area is a mosaic of different source types ranging from the Euphrates-Tigris basin and the Syrian desert, stretching to Saudi Arabia and the Sistan basin, all of which are profoundly influenced by human activity through land use change and the construction of reservoirs (Rashki et al. 2012; Zender



Fig. 2 Physical processes that lead to sand and dust emission and govern the lifetime of SDS

2003). From this area, it is estimated that the annual mean dust flux is between 40 and 500 Tg year⁻¹ (Ginoux et al. 2004; Miller et al. 2004; Tanaka and Chiba 2006).

Meteorological conditions and climate variability have both been proven to have an exert influence on SDS. Bodenheimer et al. (2019) concluded that SDS of eastern origin are mostly related to the Red Sea Trough system while, Varga et al. (2014) identified that SDS of western origin are mostly related to Mediterranean cyclones during winter and Sharav cyclones during spring. Consequently, the SDS in the region can affect the air quality in mainland Greece, the North Aegean Region, Cyprus, Israel, and Turkey (Çapraz and Deniz 2021, Triantafyllou et al. 2020; Tsiflikiotou et al. 2019; Vratolis et al. 2019, Krasnov et al. 2016a; Mouzourides et al. 2015).

As a dominant phenomenon for EMR, SDS is being recorded by researchers in different studies in terms of their frequency and intensity based on very high-resolution ground and satellite data. For example, based on ground data from stations within Pey et al. (2013) records that in the whole Mediterranean Basin, the south-eastern part which includes the EMR has higher annual dust contributions than the west or the western part of the Basin. Furthermore, over a 49-year data analysis from 1958 to 2006, Ganor et al. (2010) found an increasing trend of SDS in the EMR. In addition, individual studies in specific areas have been monitoring the trends of SDS. For Cyprus, data between 1993 and 2008 from an urban station in Nicosia and a background station in Agia Marina found that average daily concentrations of PM₁₀ during SDS have decreased from the year 2000 to 2008, even though the number of dust days has increased (Achilleos et al. 2014). In another instance, ground monitoring stations measuring PM_{10} concentrations in 3 cities of Israel between 2001 and 2015 identified more extreme concentrations PM₁₀ due to SDS in the years after 2009 (Krasnov et al. 2016b). Lastly, Aslanoğlu et al. (2022) identified the climatology of dust in the area using CALIOP and MODIS satellite systems for a 9-year period (2007–2015) in Turkey. This study found that the eastern part is more affected than the western part of the country. In general, studies in the area were conducted without uniform methodology in different areas of the EMR. An exception was found in the literature, for example, Achilleos et al. (2020) focused on 3 sites simultaneously (Cyprus, Crete-Greece, and Israel) over a 12-year period from 2006 to 2017. Their study provides a holistic approach of identifying dust days with PM ground measurements, satellite products, and meteorological conditions. They also found that there is no increasing or decreasing trend, but rather a quasi-steady state of dust days in the area for the period of study.

Due to the variability in meteorological conditions, modelling and forecasting SDS is complex and significant SDS events in the EMR are the centre of attention and study. For example, a severe SDS over the Eastern Mediterranean that took place in September 2015 (cf. Figure 3) was missed by the operational models due these weaknesses (Mamouri et al. 2016; Solomos et al. 2017). The magnitude of this event, although rare, showed how many countries in the area can be affected, highlighting the need for adequate concerted efforts for mitigation and adaptation measures. Other significant events have also been investigated in the EMR as the 'Minoan Red' event in Crete-Greece during late March of 2018 (Monteiro et al. 2022) or the intense SDS in April of 2005 over Greece (Kaskaoutis et al. 2008).

3 Impacts of SDS

3.1 Negative impacts

Hazards due to SDS have been recognised in all three phases of the phenomenon, i.e. from the entrainment-lifting of sand and dust, to transport and deposition (Middleton 2017).



Fig. 3 Satellite images of the EMR as the SDS event progressed in September 2015, extracted from NASA Earth Observing System Data and Information System (EOSDIS), https://worldview.earthdata.nasa.gov

Hazardous impacts may be identified but are not limited to human and animal health (e.g. Vodonos et al. 2014; Neophytou et al. 2013a, b), satellite signals (Solheim et al. 1999), and air flights (Middleton 2017), whereas it may have a substantial impact on the performance of renewable energy infrastructure such as photovoltaic panels (Conceição et al. 2018; Saidan et al. 2016).

In drylands, where degradation is high (Reynolds et al. 2007), deflation removes the topsoil where nutrients are mostly found, reducing the productivity of agricultural areas and damaging young plant roots (Armbrust and Retta 2000; Mrabet 2002). Furthermore, damage to the plant leaf tissue due to sandblasting affects the overall growth and performance of the plants and consequently impacts agricultural production (Stefanski and Siva-kumar 2009). Infrastructure is also affected by deflation in near-source areas, where the entrainment and transport of large quantities of particles occur (Huszar and Piper 1986; Wang et al. 2010). For example, in Kuwait, the annual cost of removing sand and dust from roads is \$9.36 million annually (Al-Hemoud et al. 2019).

The hazardous impact of SDS on human and animal health (Heal et al. 2012) increases in complexity in urban areas, where dust particles are mixed with anthropogenically emitted organic particles (Riemer et al. 2019) because of the particular airflow structures in compactly built areas and street canyons. Such airflow structures have been found to substantially enhance the accumulation of entrained particles transported over long distances and reduce the breathability capacity (Panagiotou et al. 2013) of urban streets, which can be thought of as the *lungs* of a city. Due to the reduced breathability capacity in highly compacted built areas, higher concentrations of particles are observed within street canyons than in other areas, resulting in the increased exposure of pedestrians and citizens to particulate pollution; such observations were made in laboratory experiments (e.g. Neophytou et al. 2014), computational simulations (e.g. Kubilay et al. 2017), and field measurements within urban street canyons of compacted built areas (e.g. Karra et al. 2017).

In the EMR, especially for countries such as Cyprus, Greece, and Israel, several studies demonstrated associations of PM_{10} levels during DDS outbreaks with increased total and case-specific mortality and hospital admissions for asthma and chronic obstructive pulmonary disease (Middleton et al. 2008; Neophytou et al. 2013a, b; Samoli et al. 2011, Vodonos et al. 2014).

Sand and dust particles can also negatively affect the hydrologic cycle and global radiative budget (Hui et al. 2008; Kaufman et al. 2005) because they act as cloud condensations and ice nuclei or absorb and scatter incoming solar radiation, respectively. Specifically, dust particles form longer-living shallow clouds and decrease the size of the droplets by 10–30%. As a result, they are further susceptible to evaporation by absorbed solar radiation, which in turn inhibits precipitation (Lohmann and Feichter 2005). Moreover, during the SDS deposition phase, dust particles can 'contaminate' water bodies by changing their chemical characteristics (Griffin et al. 2001).

3.2 Positive impacts

Positive effects of SDS have also been reported in the literature. Micron-sized particles (i.e. $1-20 \mu m$) significantly affect the net change in the energy balance of the earth's system. Depending on their characteristics, particles can absorb or scatter solar radiation, causing a cooling effect in the atmosphere and ocean (Ginoux 2017; Lau and Kim 2007). Studies have reported that large sand and dust particles can act as greenhouse gases by heating the atmosphere (Kok et al. 2017; Mahowald et al. 2014). Similarly, large dust particles are effective cloud condensation nuclei when they are freshly emitted, mainly because of their size, and when coated with soluble or hygroscopic material as they age in the atmosphere (Choobari et al. 2014; Kumar et al. 2011; Levin 2005).

Positive effects have also been observed because significant amounts of nutrients are transferred across regions during SDS. For example, the cross-fertilisation of the Amazon forest, where the soil is low in nutrients due to high erosion from high rainfall, and the Bodele depression in Chad acts as one of its main nutrient sources (Ben-Ami et al. 2010; Koren et al. 2006; Swap et al. 1992). In addition, plankton and chlorophyll productivity in ecosystems is enhanced because of the settling of dust in the sea (Gallisai et al. 2014). Chlorophyll in the central and eastern Mediterranean was found to be affected mostly by sand and dust deposition, particularly during spring. Finally, sand and dust can be partly alkaline and negate the effects of acid rain (Han et al. 2011).

Overall, SDS have significant positive effects on the environment and ecosystems, and consequently on humans, indicating that mitigation measures must be designed and implemented to moderate their negative effects without disturbing the ecological balance.

4 Mitigation measures in the emission source areas

Sand and dust sources have different characteristics and are therefore subject to different mitigation approaches. This section overviews the mitigation measures and assesses their success or implementation challenges near-source areas. We should note here that mitigation measures are the means used in the source areas to prevent, reduce, or control SDS. In some cases, mitigation measures also include restitution for any damage to the environment caused by the emissions of sand and dust through compensation, replacement, restoration, or any other actions.

The mitigation actions reviewed in the following subsections, which have been applied at different scales, are structured based on the type of emission sources (Fig. 4). Table 1 summarises all the mitigation measures based on their applications at different locations. Evidently, the most systematic efforts to reduce dust episodes are mainly located in Central Sahel-Niger, Mali, and Burkina Faso. Few studies have been conducted on the Arabian Peninsula, where sand and dust mitigation measures have been considered. The insufficient information on this subject in the Arabian Peninsula may be due either to the insufficient interest in mitigation measures or the limited publication of relevant activities.

4.1 Mitigation measures for wide-scale SDS

Large-scale and cross-sectoral actions in the societal, agricultural, and economic dimensions have been initiated in some source emission areas.

The massive planting of trees around the Sahel has been given a broader sense with the Great Green Wall (GGW) of the Sahara and Sahel. This initiative was first introduced in the 1980s and formally adopted in 2007 in the National Action Plans of several Sahelian countries (UNCCD 2016). This enormous plan comprises a 15-km-wide and 7775-km-long tree line from Dakar to Djibouti in order to reduce wind erosion, desertification, and ultimately SDS. As of 2020, some of the countries involved with the project have made significant progress in land restoration while others not so much. Specifically, in the UNCCD (2020) report, all 11 countries taking part in the project have reported financial challenges while governance, monitoring, and technical challenges were also identified between 2011 and 2017. Such forest plantations can fail for many reasons, e.g. from anthropogenic (e.g. lack of planning, injuries to young plants during planting, and unsuitable species of trees) and natural reasons (e.g. low rainfall and high temperatures (Jackson 1984). Thus, maintaining and monitoring a plantation in its early stages does provide with a higher survival rate. Although millions of trees have been planted along the GGW, much more planting must be performed, and the plan is expected to be completed in the next decades. Large-scale tree plantations, similar to the GGW in Africa, not only mitigate the effects of SDS but also reduce the effects of climate change by reducing the greenhouse gases in the atmosphere (Bastin et al. 2019; Benjaminsen and Hiernaux 2019).



Fig. 4 Graphical summary of mitigation measures based on emission sources

Table 1 Summary of the mitigation me	asures found for the source areas based on the	r application	
Source/location	Mitigation actions	Strategic points	References
Wide scale Sahel (Niger, Burkina Faso)	Farmer-Managed Natural Regeneration	Millions of hectares of land have been regreened Self-sufficient measures to promote stability	Reij et al. (2009) Pye-Smith (2013) Abdou (2013)
Sahel (Across Africa)	Great Green Wall for the Sahara and the Sahel	A large-scale plan to reduce desertification, wind erosion, and, ultimately, SDS	UNCCD (2016)
Sahel (Cameroon, Central African, Republic, Chad, Libya, Niger, Nigeria)	Lake Chad 2009–2017 restoration pro- gramme	Stability of dried areas and greening of land	ADF (2020)
Sahel	Soil and water protection and conservation schemes	Stabilisation of sand and restoration of areas and regreening	GIZ (2012)
Northern Africa (Morocco)	Green Morocco Plan	Harvest of rainwater and transplantation of rosemary to increase vegetation	Solh and Saxena (2011) Homrani Bakali et al. (2016)
Croplands			
Sahel	Crop residues	Significant reduction of erosion but not com- parable results Very low crop residues are sufficient for reducing wind by a factor of 4	Sterk (2003) Abdourhamane Toure et al. (2011)
Sahel (Burkina Faso)	Parkland system	Shrubs reduced wind speed and sediment transport up to 7.5 times	Leenders et al. (2007, 2011, 2016)
Sahel	Fallow band system	Can effectively capture soil particles and coarse organic matter	Ikazaki et al. (2011)
Sahel	Windbreakers	Prevents wind erosion but can cause dif- ficulty in agricultural practices	Sterk (2003)
Northern Africa (South Tunisia)	Mouldboard	Mouldboard has significantly better results in reducing wind erosion than other means	Labiadh et al. (2013)
Northern Africa (Tunisia Morocco)	No-tillage method	The no-tillage practice can decrease wind erosion and increase soil organic carbon	Ben Moussa-Machraoui et al. (2010) Mrabet et al (2012) Moussadek et al. (2014)

Table 1 (continued)			
Source/location	Mitigation actions	Strategic points	References
Northern Africa (Morocco)	Intercrop	Trees among crops can help prevent soil erosion	Daoui and Fatemi (2014)
Middle East (Northern Syria)	Zero tillage and crop residues	Conservative agriculture techniques, increased soil organic carbon and decreased wind erosion	Sommer et al. (2014)
Middle East (Northern Syria)	Crop residues	All residue treatments have a positive influ- ence on soil organic matter and improve soil properties and moisture levels	Sommer et al. (2011)
Middle East (Jordan)	Stone fragments in fields	Using 5% and 15% stone fragments in a field can reduce soil erosion by 35% and 53%	Abu-Zreig et al. (2011)
Rangelands			
Sahel (North Ethiopia)	Enclosures	Enclosures of areas for preventing grazing have produced very good restoration results	Yayneshet et al. (2009) Gebrehiwot and Veen (2014)
Northern Africa (Southern Tunisia)	Sand fences	Sand fences from palm leaves were built next to a road, to create a dune and enable sand-dust deposition	Wiggs (2011)
Northern Africa (Mauritania)	Water usage	Use of treated sewage and unadulterated seawater for sand dune stabilisation	Badescu and Cathcart (2010)
Middle East (Iran)	Water usage policies and monitoring	Measures that have contributed considerably to stabilising the water level	ULRP (2018) Danesh-Yazdi and Ataie-Ashtiani (2019)
Middle East (Israel)	No disturbance	Microphytic communities in the northern Negev Desert form a biological crust that further enhance the stabilisation	Veste et al. (2001)
Industrial setting/mines			
Not found	I	1	1

To reverse the adverse effects of significant losses of fertile land due to desertification and population pressures, the 'Maradi Commitment' was established in Niger in the 1980s, wherein a block of 10 ha in each city was planted with trees (Larwanou et al. 2006). Additionally, farmers were urged to use the trees naturally growing in their fields as soil stabilisers and as a source of fuel and food for livestock through continuous harvest techniques under a programme called Farmer-Managed Natural Regeneration (Abdou 2013). Furthermore, the government of Niger adopted the 3N initiative, *Nigeriens Nourishing Nigeriens*, to promote sustainability and combat poverty, land degradation, and hunger (Abdou 2013). Today, villages have 10 to 20 times more trees than at the beginning of the programme, and 5 million ha of land is estimated to have been regreened (Pye-Smith 2013; Reij et al. 2009).

Degrading lake systems and drying rivers can result in the exposure of soft sediments, as in the case of the Bodele depression in the Sahel region, which is one of the biggest sources of sand and dust. The lake was part of Mega Lake, Chad, which once covered 350,000 km² (UNESCO 2004). Since 1950, the lake went through different dry states up to 300 km² in the 1980s, exposing much of its highly erodible sediment and degrading nearby agricultural lands and pastures because of climate variability and human intervention (Galeazzi et al. 2017; LCBC 2016; Mahmood and Jia 2019). Millions of individuals reside in and depend on the area; therefore, in 2009, measures were implemented through the Lake Chad Basin Sustainable Development Programme to reverse the degradation situation. At the end of the programme in 2017, local plants covering 148 ha stabilised the dunes, and new agricultural and pastoral lands were developed (ADF 2020). Notably, other soil and water protection programmes and conservation schemes adopted in the Sahel aimed to stabilise the blowing sand from dunes, fence off areas that have been restored, and plant trees. These aims have been successfully executed due to German cooperation and funding from African projects (GIZ 2012).

In Northern Africa, the southeast rangelands of Morocco were restored by harvesting rainwater and transplanting rosemary, which increased vegetation, serving the needs of the local population (Homrani Bakali et al. 2016). This action was part of the Greater Green Morocco Plan that started in 2008, which attempted to increase agricultural production and the income of farmers as well as to ensure sustainable development (Solh and Saxena 2011).

The area with the least amount of intervention was the Middle East. The only relevant reference found in the literature review was the signing of a memorandum agreement between Iran and Iraq in 2011 (Ganjalinejhad et al. 2018) to facilitate the management and control 1 million ha of deserted land, but whether the plan was implemented was unclear.

4.2 Cropland

Croplands provide another significant source of SDS, and actions are implemented by farmers and scientists to reduce soil erosion and increase crop yield. There are many cultivated land areas in the source regions of the Sahel, Northern Africa, and the Middle East (Ginoux et al. 2012). The incorrect or limited use of best practices to prevent soil erosion can have many disadvantages, as highlighted by Govers et al. (2017). In the source areas investigated, the soil condition was poor, and the trend of soil erosion was increasing, making them susceptible to desertification (FAO and ITTPS 2015).

Soil management techniques and barriers have been used for reducing the loss of nutrient-rich topsoil in croplands. Bielders et al. (2001) examined the use of ridges

perpendicular to the prevailing wind direction, showing a drastic decrease in soil erosion by 57%, and that reducing the spacing between the ridges increased their effectiveness in an on-farm experiment in Southwest Niger. However, this practice had a short life span due to rainfall and thus requires continuous maintenance, especially in the sandy soil types of the Sahel (Fryrear 1984).

Mulching is another countermeasure to wind erosion and was first implemented in Niger. Nigeriens used manure and crop residue to cover sandy soil. However, due to the different methodologies used in the studies, their mitigation effects could not be accurately determined (Sterk 2003). To evaluate the effectiveness of this measure, Abdourhamane Toure et al. (2011) investigated the quantity of crop residue required to control wind erosion by monitoring the erosion flux over 2 years. The results showed that generally low crop residues of approximately 100 kg ha⁻¹ or approximately 2% of the total cover are sufficient for reducing wind erosion by a factor of 4.

In another practice, parkland systems in Burkina Faso (i.e. landscapes where scattered vegetation of trees and shrubs can grow) are sustained in cultivated or recently fallowed fields (Leenders et al. 2016). Shrubs were found to reduce wind speed and sediment transport by up to 7.5 times (Leenders et al. 2007; 2011). Intercropping, parkland-style vegetation with trees among crops, as practiced in Morocco can also help reduce soil erosion, as indicated by Daoui and Fatemi (2014). In contrast, changing rangeland to agricultural land can increase wind erosion and contributed to a net soil loss of 62 Mg/ha per year, which is five times more than the tolerable soil loss rate for the area (Houyou et al. 2016). Thus, exploitation of land for crop cultivation can lead to increased erosion if appropriate measures are not implemented to mitigate it.

The use of trees as windbreaks for crop protection in croplands is a common practice, and they can also act as soil stabilisers. The fallow band system, which was also applied along the Sahel, comprised herbaceous windbreaks that were 5-m wide and were placed against the erosive wind direction from the rainy to the dry season. This fallow band system has shown that over 3 years, it can effectively capture soil particles and coarse organic matter (Ikazaki et al. 2011). Tree-windbreakers have also been used in the Sahel, and Sterk (2003) indicated that trees can inhibit wind erosion. However, several limitations have been observed, such as the competition of trees for water and nutrients with the crops, solar radiation, cattle grazing, and land tenure concerns.

In addition to the aforementioned passive measures, agricultural machinery can play an influential role in soil stability. An experiment was performed in South Tunisia to assess how the use of different machinery tillage approaches affects soil erosion. The findings demonstrated that using a mouldboard attached to a tractor in cultivated fields had significantly better results in reducing wind erosion than using tractor discs or tillers as part of conventional tillage techniques (Labiadh et al. 2013). Additionally, in Morocco, decreased wind and water erosion was observed when minimum disturbance equipment, no-tillage methods, no-till disc drills, and dry seeding methods were used as part of the conservation tillage techniques (Mrabet et al. 2012). Generally, the complete removal of residues reduces crop yield because of high erosion (Mrabet 2002). Furthermore, Bot and Benites (2005) reported that increased soil organic matter and erosion control can be achieved. Ben Moussa-Machraoui et al. (2010) showed that under different crop types, the soil organic carbon and matter were enhanced when no-tillage practices were used, compared with their levels while using conventional tillage in a 4-year study in northern and north-western Tunisia. These results agree with the findings of Moussadek et al. (2014) in Central Morocco, where the use of no-tillage practices for 5 years increased soil organic matter from 2 to 10% in two major soils, vertisol and cambisol. Likewise, in the Middle East,

and specifically in Northern Syria, zero tillage and crop residue measures were applied from 2008 to 2012 in different experimental crop fields to assess the difference between conventional and conservative tillage. The results showed that in conservative agriculture, high soil organic carbon and microbial biomass were observed, showing the potential for decreasing erosion (Sommer et al. 2014). Moreover, all residue treatments had a positive influence on soil organic matter and improved soil properties and moisture levels (Sommer et al. 2011). An additional example of soil loss reduction is the use of stone fragments in Jordan. Abu-Zreig et al. (2011) found that using 5% and 15% stone fragments in a field could reduce soil erosion by 35% and 53%, respectively. Some of the practices used for countering soil loss are also helping with crop yield, if used correctly. Stone fragments are found to increase crop yield in various cases and hold moisture in the soil, while they help with surface runoff and erosion (Nyssen et al. 2006; Katra et al. 2007). Considering the area of application, a mitigation measure can have the opposite effects if not applied properly.

4.3 Rangelands and natural ecosystems

Addressing the changes to rangelands and natural ecosystems, which may change SDS because of anthropogenic and natural intervention, through various measures is essential. One third of humanity lives in rangelands and drylands where semi-arid or arid climates persevere, covering more than 40% of the land area. Consequently, their stability is crucial for human survival (Reid et al. 2014; Safriel et al. 2005; Middleton and Thomas 1997). Moderate grazing regimes in a nomadic style are likely to improve rangeland regeneration rather than ranching or stocking rates, according to a long-term study of monitoring vegetation between 1981 and 2007 in the central Sahel (Miehe et al. 2010). In North Ethiopia, enclosure strategies to prevent grazing have produced very good restoration results (Gebrehiwot and Veen 2014; Yayneshet et al. 2009).

Furthermore, 31% of SDS on a global level are attributed to hydrological sources, such as river valleys, alluvial fans, and playas (dried lakes) (Ginoux et al. 2012; Goudie 2018). In Iran, Lake Urmia is an example of the anthropogenic influence on water bodies and their desiccation: the lake was severely downsized because of overconsumption and the building of dams, reducing it to 2366 km² by 2011 (Ouria and Sevinc 2016; Garousi et al. 2013). With 88% of its water cover lost from approximately 2000 (AghaKouchak et al. 2015), an estimation of a 30–60% increase in PM₁₀ in nearby cities was recorded (Sotoudeheian et al. 2016). To counteract this decline, the Urmia Lake Restoration National Committee was established in 2013, to devise and implement a plan to stabilise the decline and restore the lake through a series of measures. These measures comprised a decrease in agricultural land and, consequently, the use of agricultural water, additional water release from the dams to increase the water level of river valleys, and the control of the illegal withdrawal of water from the lake (ULRP 2018). In 2016, the water level was stabilised, and other actions, such as replacing the old irrigation systems with new, efficient versions, showed that the Urmia Lake Restoration plan has contributed considerably (Danesh-Yazdi and Ataie-Ashtiani 2019). However, Zucca et al. (2021) indicate that many of the reported mitigation measures of SDS impacts from hydrological sources are just mentioned as possible solutions based on statistical or other observations without any evidence of successfulness. In the review, a list of Sustainable Land and Water Managements (SLWM) practices are identified based on their nature of application, intervention zone either upstream,

playa or downwind, and importance, with the help of the World Overview of Conservation Approaches and Technologies. Examples of high importance are integrated watershed management planning and implementation as an approach for all 3 intervention zones, improved water use efficiency as technologies related to water primarily for upstream and playa zones, and windbreaks as technologies related to wind and water erosion for playa and downstream zones.

Sand dunes cover large parts of natural ecosystems, such as deserts, and their formation is controlled by sand supply, wind regimes, and vegetation (Lancaster 2011). Significant shifting patterns in sand dunes have been associated with the topography and regional atmospheric circulation of an area (Washington et al. 2006). Furthermore, in arid areas, the chemical composition of sand dunes plays a significant role in wind erosion reduction; for example, the Negev and Sinai deserts contain 17% silt and clay, enabling them to be stabilised naturally (Tsoar and Zohar 1985). The microphytic communities in the northern Negev Desert form a biological crust that enhances stabilisation (Veste et al. 2001), thus not disturbing the areas can be crucial.

Additionally, anthropogenic interventions, such as the formation of sand fences from palm trees next to a road in southern Tunisia (Wiggs 2011) and the use of unadulterated seawater and cleaned urban sewage in Mauritania (Badescu and Cathcart 2010) to stabilise sand dunes, constitute sustainable practices to reduce wind erosion in natural ecosystems such as deserts.

4.4 Industrial settings-mineral operations

Mitigation measures are not only those directly related to natural sources and ecosystems but also those applied to sources related to large-scale anthropogenic activities such as mining and industrial operations. Fugitive dust from mining or industrial operations can be more harmful than normal dust emissions from deserts because it can contain contaminants from chemical additives used for mineral extraction processes and heavy metals (Csavina et al. 2011, 2012). An example is arsenic: 40% is emitted from smelting operations (Alloway et al. 1997). Dust particles from mining operations have been found to contribute significantly to suspended particle concentrations, 25% of which are respirable (Zota et al. 2009) and are mainly emitted from haul roads, tailing beaches, and slope areas (Park et al. 2019).

Despite the reports on emissions from mines in the dust source areas examined, such as in Mali and Niger (Garrison et al. 2014), Iran (Monjezi et al. 2009), and Algeria (Mokadem et al. 2014), no mitigation measures have been reported. However, such measures have been applied (e.g. Park et al. 2019) in areas not covered by this review. Nevertheless, they are briefly cited as examples or recommendations as follows:

- i. Water spraying on tailings
- ii. Covering tailings with other materials
- iii. Stabilising tailings with vegetation
- iv. Stabilising tailings with chemicals

The use of chemicals (e.g. biopolymers for stabilisation) was found to be more effective in enhancing moisture retention than using only water spray on tailings (Chen et al. 2015). Furthermore, phytoremediation, used in the USA as a measure of controlling dust emissions from tailings, showed a decrease in dust particle emissions and was comparable to the emissions from undisturbed grasslands (Gil-Loaiza et al. 2018).

5 Adaptation measures

Except for large-scale forecasting, covering the entire EMR region, we found no other wide-scale adaptation measure in the literature. On these grounds, small-scale and personal adaptation measures are recommended.

5.1 Monitoring and forecasting of SDS

An intermediate step for the implementation of other adaptation measures, highlighting the necessity and usefulness of monitoring and early prediction of SDS in areas far from emission sources, is important. Monitoring of SDS has always been conducted using long-term sand and dust observations and meteorology (Goudie and Middleton 1992). Before the advancement of atmospheric modelling and remote sensing, the categorisation of SDS severity was established based on the intensity of wind speed and reduction of visibility (Joseph et al. 1980; Xin-fa et al. 2001).

However, current technological and scientific progress has enabled scientists to monitor and predict SDS. The monitoring of sand and dust was recorded mainly by the ground stations. Networks such as the European Monitoring and Evaluation Programme include background stations that are located far from anthropogenic sources to record pollutants from long-range distances. Continuous monitoring is conducted at each site by using high temporal resolution equipment, such as a tapered element oscillating microbalance. These monitoring techniques are useful for evaluating models worldwide (Eleftheriou et al. 2021). AERONET, which employs a network of sun photometers (Holben et al. 1998), is another example of an observational system. Orbital or geostationary remote sensing platforms such as MODIS, SEVIRI, and CALIPSO also provide many satellite products with atmospheric aerosol properties at various temporal and spatial resolutions (Paz et al. 2013). Despite their great capabilities, monitoring techniques have limitations because most of them cannot provide the source apportionment of the particles in real time. In addition, sun photometers and lidars cannot perform reliable measurements during intense dust events (Fernández et al. 2019), and providing reliable satellite products is difficult when the atmosphere is cloudy, smoky, or dark.

Additionally, regional-scale forecasting of SDS requires the modelling of complicated physicochemical processes that occur in the environment. This finding implies methodological variability (various methods and databases, i.e. ground observations, satellite products, weather, and dust models), different levels of complexity, and ultimately significant uncertainty in forecasting. This uncertainty is due to many reasons, such as the lack of sufficient evaluation measurements appropriate to enable the confirmation of different stages (e.g. emission and deposition schemes) in the modelling processes, sub-grid variation causing differentiation in the friction velocity, particle size distribution, and the development of satellite methodologies (Shao and Dong 2006). Moreover, due to the use of 'static' land use and cover databases of SDS at sources, seasonal variations in land are not considered. In recent years, attempts have been made to produce dynamic SDS source databases (Kim et al. 2013; Solomos et al. 2019; Vukovic et al. 2014).

Such large-scale forecasting data is being hosted on online databases such the Sand and Dust Storm Warning Advisory and assessment System (SDS-WAS) which was established by the WMO in 2007 and is hosted in Spain for the European, African, and Middle East regions and in China for the Asia/Central-Pacific region (SDS-WAS 2020). Nowadays, a new updated platform is used the WMO Barcelona Dust Regional Center (https://dust.aemet.es/) and this centre is responsible for the research activities and operations of the WMO related to SDS. This online database hosts many, if not all, of the available operational dust models which are updated on a daily basis. Each individual model is run by a different country/authority/institution with different input data and modelling processes, resulting in varying outputs of dust concentration on the same area of interest.

In a long-term intercomparison of models with satellite products, substantial discrepancies have been recorded among several global- or regional-scale models (Evan et al. 2014). The lack of observations of soil characteristics (e.g. size and mineralogy) and gaps in meteorological conditions at source areas, which are important for modelling SDS, can result in differences among models (Benedetti et al. 2018; Roberts et al. 2015; Menut 2008).

Overall, monitoring and forecasting tools are used and can be used as adaptation measures to provide necessary warnings for individuals to be protected from incoming SDS, especially SDS. However, as reported in the literature, many concerns must be resolved to obtain the most timely, correct identification of SDS and their severity.

5.2 Personal exposure and protection and the reliance on early warning systems for SDS

Historically, SDS have not been considered harmful to humans, because of their natural origin and crustal composition. In line with this approach, legislation of the European Union considers SDS impossible to prevent, implicitly harmless, and discounts their contribution to the daily and annual air quality standards of PM_{10} . However, during the last two decades, several epidemiological studies from the EMR countries, such as Cyprus (Middleton et al. 2008; Neophytou et al. 2013a, b), Greece (Samoli et al. 2011), and Israel (Vodonos et al. 2014), have demonstrated associations of PM_{10} during SDS outbreaks with increased total and case-specific mortality and hospital admissions for cardiovascular diseases, asthma, and chronic obstructive pulmonary disease.

Currently, during SDS, competent authorities and mass media in the countries of the region issue non-standardised warnings to the general public and vulnerable groups, most commonly advising them to remain indoors and reduce outdoor activities. There is limited evidence on the societal concerns and risk perceptions for SDS and the health-related effects (Im et al. 2006). Only few studies examined the current knowledge and relevant practices of competent authorities or the knowledge and perceptions of involved social stakeholders in the region regarding the health effects of SDS (Kinni et al. 2021).

There is no scientific evidence on the efficacy of any of the recommendations in reducing exposure to SDS PMs or mitigating related health effects. The main goal of the demonstration project 'MEDEA' (Mitigating the Health Effects of Desert Dust Storms Using Exposure-Reduction Approaches; https://www.life-medea.eu) is to provide field-based evidence for the feasibility and effectiveness of adaptation measures to SDS in South-Eastern Europe, focusing on exposure reduction approaches. In this project, which is ongoing, exposure reduction is pursued by implementing both outdoor and indoor interventions (Kouis et al. 2021). For the outdoor intervention, children with asthma after receiving timely personal alerts and specific recommendations

by smartphone and web-based applications and tools demonstrated significantly reduced outdoor exposure during SDS events by reducing time spent outdoors (-28%) and reducing outdoor physical activity (-13%) (Kouis et al. 2023). For the indoor intervention, children with asthma and adults with heart arrhythmias were asked to close windows and doors, seal possible cracks around windows and doors to minimise home ventilation, and use an air cleaner continuously to filter indoor air and managed to significantly reduce indoor PM_{2.5} and PM₁₀ by 43% and 41%, respectively, in comparison to the control group (Achilleos et al. 2023). In addition, anecdotal evidence from these studies, which is currently in the publication process, documented significant improvements in asthma symptom control in children and health-related quality of life in adults with heart arrhythmias. Implementation of interventions also depends on validated models for forecasting SDS in EMR, as well as early dissemination of warnings and audio-visual recommendations for exposure reduction via a web-based platform and smart mobile application to alert patients of upcoming SDS.

For the EMR, an early warning system, the SDS-WAS platform and now the WMO Barcelona Dust Regional Center as mentioned before, is commonly used. The concentrated expertise and data found on these online databases has proved valuable for competent authorities around the world to issue warnings for the public. Typically, the multi-model median (MMM) is being taken as an output or individual model based on each competent authority's experience or statistical analysis.

Exposure-reduction strategies have an important role in adaptation strategies. The employment of novel environmental epidemiology and telemedicine methods to assess personal compliance to recommendations, measure exposure to air pollutants in indoor and outdoor environments, and monitor clinical outcomes among the population is very important in the evaluation of these strategies. The methods previously employed in air pollution health effect studies have several inherent inaccuracies in assessing exposure and health outcomes. Exposure estimates are typically based on measurements conducted at sparsely distributed monitoring stations. These approaches use outdoor air pollution concentrations as a proxy for total exposure; therefore, they lack information on indoor air pollution levels, which introduces significant exposure errors (Baek et al. 1997; Hoek 2017; Leung 2015). Furthermore, exposure estimates are usually assessed for a given residential address, without considering participants' activity and mobility throughout the day (Park and Kwan 2017; Setton et al. 2011; Wu et al. 2011).

With modern technologies, wearable GPS and activity sensors can be used to continuously measure activity and time spent indoors and outdoors. This approach enables the assignment of exposure to the respective air pollutant levels measured in indoor and outdoor environments, differentiated by activity levels throughout the day, providing a high spatiotemporal resolution (Yarza et al. 2020). The health effects of desert dust exposure are typically assessed using ecological retrospective data on major outcomes such as death, hospital admissions, and outpatient clinic visits (Vodonos et al. 2014; Neophytou et al. 2013a, b; Samoli et al. 2011; Middleton et al. 2008). However, data on hospital admissions and outpatient clinic visits are influenced by subjective health care seeking behaviour and are therefore problematic in evaluating the onset, duration, and severity of an outcome (Yarza et al. 2020). With telemedicine methodologies, a range of clinical outcomes may be assessed beyond what standard tools, such as validated clinical symptoms questionnaires offer. In this context, health parameters such as heart rate, blood pressure, and oxygen saturation can be obtained from wearable sensors, and heart arrhythmias can be obtained from implanted pacemakers, providing continuous measures of a whole range of clinically relevant parameters that can be related to exposure to SDS.

5.3 Green infrastructure (GI)

The literature suggests a few GI-related adaptation measures for SDS in areas away from the sources. Therefore, adopting ideas and solutions that have been successfully implemented for similar environmental topics such as traffic pollution is necessary. GI, also referred to as nature-based solutions (NBSs; Debele et al. 2021), is a passive control measure, such as planting trees hedges, or shrubs to reduce air pollution (Abhijith and Kumar 2020; Kumar et al. 2019, 2022). For example, Abhijith et al. (2017) reviewed approaches and methodologies applied to open roads and street canyons and highlighted their benefits. For instance, trees in canyons with low porosity and high height can act as walls, trapping the pollution underneath them, forcing the air to flow above them, and thereby reducing air exchange (Abhijith and Gokhale 2015; Janhäll 2015). Therefore, trees can also be used to filter ambient air from dust or resuspended particles (Ottosen and Kumar 2020). Planting trees in compact urban areas to reduce air pollution and increase human welfare is not an easy task because many parameters need to be considered before committing to this action (Kumar et al. 2019; 2021).

Crucial parameters for consideration include, for example, the tree type, geometry of the area to be planted, and spacing between the trees (Barwise and Kumar 2020). Moreover, hair and waxes on certain types of trees (Sæbø et al. 2012), and the surface of leaves and their shape (Perini et al. 2017), influence the number of particles deposited on them, thereby increasing the accumulation of pollutants. Thus, GI can be of importance in adaptation measures because it can allow the deposition the atmosphere PM on them and mitigate against natural hazards such as flooding, landslides, and heatwaves (Debele et al. 2021; Kumar et al. 2020). The implementation of this measure, especially in an urban environment where the air quality is heavily burdened and buildings are much denser, must be carefully considered.

6 Social, economic, and political dimensions of mitigation and adaptation measures

Examples of how challenging the implementation of SDS mitigation and adaptation practices is can be realised when the socioeconomic and political aspects of each area are considered. For example, in most SDS source areas, the rural population is among the poorest worldwide, and there are many environmental problems such as land degradation, soil erosion, and the lack of water (Middleton et al. 2011).

Land degradation in drylands is caused by a number of natural and human-driven activities such as extreme climatic variations and overgrazing respectively (Davies et al. 2012; IUCN 2017). Grazing has been found in literature (Rowntree et al. 2004) to be of a complex desertification driver, as it is at the same time interacting with other factors affecting the area. These factors range from the perception of how beneficial grazing is for a country (Middleton 2018), extreme weather conditions and long-lasting droughts in drylands, to human activities such as the abandonment of cultivated land and tree cutting (Dregne 2002). Although overgrazing has been identified as a cause for desertification, pastoralism for example in the Sahel is an important occupation and composes a large sum of some countries' gross domestic product (Middleton et al. 2011). Thus, the control of overgrazing and its impacts promoting desertification needs to be enforced by the understanding and considering all the other accompanying factors described above for a holistic establishment of the mitigation measures in this instance. As Butt (2016) suggested, a more in-depth investigation of the impact of overgrazing than those in the literature must be conducted because pastoralists are being pushed into additional contained areas, which might lead to overgrazing. In addition, Beyene (2010) indicated that rangeland enclosures affect pastoralists' livelihoods and do not promote sustainability. Thus, the need for grazing control of livestock in such areas is in contrast with the need for individuals to survive with measures implemented to counter the overgrazing impact. Sahelian herders also have different characteristics based on being nomadic or sedentary; thus, their characteristics differ by when, where, and how their herds are led (Young et al. 2019).

The challenge of variability of herding practices can be faced through improved communication between stakeholders for the correct and sustainable exploitation of rangelands, which seems to have increased in the 2010s. However, the lack of law enforcement and trust between herders are examples of the factors that make it challenging to promote sustainability (Coppock et al. 2017; Turner et al. 2014). Opposed to overgrazing and degradation of land in the Sahel, actions have been put in place, and ground observations and reports have confirmed the increase in vegetation, but long-term assessments indicate that biodiversity is decreasing, which might affect the environment and livelihoods (Fensholt et al. 2017). In another instance, due to the introduction of commercial agriculture in the Sahel in the 1700s, dust deposition in alluvial systems has increased dramatically (Mulitza et al. 2010), indicating that the economic and social development of the area has resulted in the formation of an SDS source. This case is similar to that of the Senegal River, which provides more material to be blown away during periods of drought, showing anthropogenic influence.

Another implementation of mitigation measures through collective action was the successful application of the Lake Chad Action Plan, but major constraints were encountered that interrupted the smooth completion of the project (ADF 2020). Such constraints were the lack of competent programme experts, leading to the constant change in plans; the weak capacity shown by governmental services in performing tasks and supervision; the low organisation of surrounding communities in applying the measures; and insecurity. The insecurity in the Lake Chad area, linked to the water scarcity caused by the drought, is directly related to the socio-political instability of the surrounding countries, which drives the area into water conflicts and increases terrorism and insurgency, such as that of Boko Haram (Okpara et al. 2015). In addition, prolonged conflicts in the Middle East and Northern Africa can result in the deterioration of land and land use changes that can influence dust emissions (Jaafar and Woertz 2016). Similar challenges were also identified in GGW project, where challenges in governance are found due to the lack of environmentally friendly political agenda and lack of coordination between authorities and relevant sectors. Additionally, the absence of monitoring and evaluation data on the ground due to the lack of finance or the overwhelming work on governments creates lack of credibility towards investors, making it harder for new funding.

Moreover, in Libya, the change from agriculture to the oil industry after the discovery of oil provided a major flow of money in the economy, which funded many projects for the development of agriculture from the 1970s to the 1980s (Sachs and Warner 1999). Medium-term plans for agriculture were implemented with great success, but with the decrease in oil prices, annual plans were established with limited success, showing the influence of the economy on the successful implementation of plans (Allafi 2014). Additionally, the role of governance in securing sustainability is crucial in Northern Africa, as proposed by Ates et al. (2016); this is because the short-lived and sometimes harmful environmental practices remain incomplete. Thus, sound governmental input must overall be established in all cases. Furthermore, national plans such as the Green Morocco Plan, which attempted to ensure sustainability in rural territories, have failed to achieve their goals (Faysse 2015) because the information used as a baseline was inaccurate.

In the Middle East, Iran's Restoration Plan for Lake Urmia cannot be considered completely successful in terms of its goals, due to lack of data and deep understanding of the various physical processes of the anthropogenic and natural stressors of the lake. Specifically, data scarcity for certain parameters, such as land cover distribution, topography, and evapotranspiration, and those physical processes that have not been adequately understood, constituted some of the reasons that the restoration plan timetable of Lake Urmia was out of schedule (Danesh-Yazdi and Ataie-Ashtiani 2019). Furthermore, Salimi et al. (2019) indicated that the implementation of the restoration phase poses great challenges due to the contradictory national policy of agricultural development and the restoration of the lake reducing water consumption, the lack of proper funding, the weak local communities that need to change their water usage norms, and the future political instability of the country that might not prioritise this project.

External financing of projects having a great positive impact against land degradation and sustainable land managements is necessary as mitigation actions in areas with high poverty and scarcity of national funding as in the case of the GGW. The need for a certain level of global international governance is becoming more evident, and in the recent United Nations Environment Programme (UNEP) summit on Climate Change (2022) in Sharm El-Sheikh Egypt between 6 and 18th of November, the possibility of funding of a global/ regional nature is being further introduced and established. The UNCCD (2020) progress report indicates that between 2011 and 2019, there is a notable discrepancy between external/international funding and national funding. Furthermore, there were a total of 870.3 million USD out of the 1.2 billion USD Sahel and West Africa Program (SAWAP) from international donors in support of the GGW project. This indicates how depended are SDS source countries on international attention and funding to proceed with successful plans. On the other hand, adaptation measures can be financed by national resources with early response mechanisms, green infrastructure, and training of the vulnerable groups with of the country that is affected through technology, making them more easily applicable and better monitored.

Regarding adaptation measures in the EMR, we found no literature on how effective the measures taken during SDS are. It is not clear whether these recommendations and warnings are established by employers or ignored because of lack of knowledge or lack of supervision from competent authorities. A recent study assessing interventions in reducing exposure to SDS indicated that compliance with adaptation measures is also dependent on behaviour, especially for children, and in school or at home (Kouis et al. 2021). Furthermore, NBS is based on co-approaches, that is, the cooperation of stakeholders at all levels and the disciplines involved in implementing them (Woods-Ballard et al. 2015). However, this interaction can reveal the perceptions of stakeholders, such as government bodies or landowners, which can cause major delays in the procedure, and their acceptance of the plan (Santoro et al. 2019).

The many conflicting interests with environmental goals have also been reported to be a problem in implementing NBS projects because developments in certain areas can be good for the short-term economy but bad for the long-term environment (Kabisch et al. 2016; Waylen et al. 2014). A common social dimension that must be considered when implementing measures is trust among stakeholders. As van Ham and Klimmek (2017) indicate, trust can be something that has been built or unbuilt during projects; thus, trust can be a decisive condition between success and failure.

In summary, in the implementation of mitigation and adaptation measures, the standard of living and culture of the affected individuals should be taken into important consideration. Values such as transparent governance, security, trust, and human welfare play an important role in the actual effectiveness of the implementation of measures. In addition, the scientific community also needs to provide a simpler context, because the context followed to date has been one of the reasons for the failures in measures' implementations (Dong et al. 2017).

7 Future pressures and approaches

Known or unknown future pressures are inevitable because of the changing norms of today, and new challenges will arise for leaders of nations and the world to manage. These pressures extend from climatic, socioeconomic, and humanitarian crises.

Simulations of climate change in source areas affecting the EMR show different results for changing temperatures and weather conditions; thus, different scenarios were investigated. The Mediterranean and West Asia will become drier, and the simulations of the Sahel will be inconsistent. Drier conditions on the ground could lead to lower monsoonal rainfall and thus longer droughts in the Sahel (Solmon et al. 2008; Yoshioka et al. 2007), which may further intensify SDS. From another perspective, a model experiment reducing SDS led to the enhancement of Saharan vegetation due to the extension of the West Africa Monsoon to the north (Pausata et al. 2016). Furthermore, observations have shown increased precipitation in the Sahel because of shorter, more frequent dry spells that can help the area's agriculture and resilience from droughts (Bichet and Diedhiou 2018). However, an increase in extreme storms has also been observed (Taylor et al. 2017); thus, a greater chance of SDS as convective storms is one of the generation mechanisms. The unpredictability of future climatic conditions must be considered when GI is used as an adaptation measure, because the increase in extreme events, such as heatwaves, droughts, and floods, can affect the selected vegetation used for air pollution protection (Bouwer et al. 2010; Forzieri et al. 2016).

Furthermore, with the increase of water needs as derived from the climate change simulations (UNEP et al. 2016), droughts may lead to conflicts (Feitelson and Tubi 2017). Upper riparian countries, such as Turkey, control the source of water for Tigris and Euphrates rivers, and without any agreements for water supply with the downstream countries, economic instability, migration, and insecurity are only a few of the problems that arise in such unstable areas (Al-Ansari 2016). Conflicts and hostile relationship between neighbouring countries sharing same hydrological resources can promote drought that controls politics and social behaviour, which among others burden the environment with additional sand and dust emissions.

Moreover, almost all countries worldwide have ageing populations because of declining fertility and increasing life expectancy (UN 2020). Related to climate change and the increase in extreme weather events (e.g. heatwaves), studies have shown that elderly individuals are more vulnerable to these conditions, increasing mortality (Chen et al. 2020; Hong et al. 2019). Additionally, deteriorating air quality, together with the potential increase in SDS due to climate change, can have an added burden on the ageing population (Simoni et al. 2015). Thus, competent authorities of affected countries need to be ready with their respective protection mechanisms, infrastructure, and organisation to adopt adaptation measures for SDS. Last, Geist and Lambin (2004) in their meta-analysis on what drivers affect desertification found that in 73% of cases, a population increase and immigration of farmers towards rangelands or herders to marginal sites are the driving issue. For example, in Lake Chad, many individuals migrate to the area every season, increasing the pressure on the environment (Zieba et al. 2017).

Focusing on the future, the Sendai Framework for Disaster Risk Reduction 2015–2030 is an important framework. Adopted in 2015 at the Sendai, Japan World United Nation Conference, this framework gives emphasis on the disaster risk management aiming at 'the substantial reduction of disaster risk and losses in lives, livelihoods and health and in the economic, physical, social, cultural and environmental assets of persons, businesses, communities and countries' as stated in UN (2015). Furthermore, there is an effort to increase the number of countries with risk reduction strategies and availability to early warning systems and to enhance international cooperation. There are 4 priorities in adapting the Sendai Framework which must be adopted at a local, regional, and global level in order to enhance governance strategies, preparedness, and coordination on disasters:

- i. Priority 1: Understanding disaster risk
- ii. Priority 2: Strengthening disaster risk governance to manage disaster risk
- iii. Priority 3: Investing in disaster risk reduction for resilience
- iv. Priority 4: Enhancing disaster preparedness for effective response and to 'Build Back Better' in recovery, rehabilitation, and reconstruction

The framework also promotes, especially for developing countries, their cooperation with international stakeholders, such as the World Bank and other international institutions in order to transfer their technological expertise, management techniques, and financial support to successfully achieve the implementation of the measures needed for.

For SDS, the United Nations Economic Commission for Asia and the Pacific (UNES-CAP) and Asian and Pacific Centre for the Development of Disaster Information Management (APDIM) have assembled guidelines based on the Sendai Framework. This initiative resulted in a step-by-step guide for the member states to enhance the states' capabilities to monitor and report data from the impacts of SDS and building an evidence base on the negative impacts (UNESCAP-APDIM 2020). Using this framework before any mitigation or adaptation measures could be an effective way of assessing the situation of how well the impacts from SDS are monitored and recorded so that the correct actions can be taken.

8 Conclusions

SDS is a phenomenon that starts locally and can grow globally. As a result, it is of increasing importance, especially in view of climate change, due to the scale of its impacts. To mitigate or adapt the effects of SDS, this critical review examines a wide range of measures and challenges in their implementation. In addition, remarks were made regarding the connection between the EMR and source areas. The main concluding remarks of this review are as follows:

(1) Mitigation measures implemented on a small scale in the source areas did not mitigate SDS on an impactful level. Large-scale implementation measures need to be introduced

to prevent or reduce sand and dust suspension from the source areas, considering the successes and difficulties of the considered measures to be applied, as in the case of the Great Green Wall and Lake Chad Action Plan. The large-scale planning requires substantial funding, especially in the SDS source areas that do not have the national resources to cover the cost of the management techniques in vast areas. This comes down largely to external/international funding such as from the World Bank and the Green Environment Facility. Except financial support in SDS source areas, technical and governance assistance needs to be also provided for the project to be successful monitoring and reporting.

- (2) The implementation of adaptation measures focuses on early warning systems using forecasting models. The models require more in situ data from the source areas, and an improved network for sharing information to obtain a more complete picture of the SDS. No other adaptation measures were found to be implemented for the EMR, or whether an extensive network of warnings for employers or businesses from authorities was successfully applied. To our knowledge, a project, the MEDEA project (LIFE + MEDEA 2020), targets its efforts on the feasibility and effectiveness of different adaptation measures for personal protection. The results of this project indicated that with use of technology and personal alerts, target groups were asked to take measures during an SDS event and significantly reduced their exposure to PM pollutants. The adaptation measures must also ensure that all SDS, independent of scale or intensity are considered. This consideration is crucial because the influence of dust (e.g. performance of photovoltaic system, contaminate water bodies) is always persistent, not only when certain political threshold criteria are fulfilled.
- (3) Assessing the socio-political-economic situation in each area is critical and must be considered in an overall approach to implementing mitigation and adaptation measures while considering the positive and negative of SDS. Despite their negative impacts, SDS can be crucial drivers of positive global impacts, such as the fertilisation of the Amazon forest. Thus, substantially reducing SDS through a series of measures can damage the environment. Examples that have affected the unsuccessful implementation of measures are the insufficient training of local populations, the lack of common goals in governmental policies, the lack of supervision in the implementation of measures, and a suitable behaviour or perception of a situation. Insecurity and the lack of funds in related projects are also situations found to be important for adopting measures for SDS.
- (4) Future pressures are expected to modify the extent of the source areas. Climate change affects differently each area of interest we investigated in this review, and the projections are inconsistent, showing, for example, either increasing or decreasing rainfall. Preventing the further impacts of climate change or stabilising it can help with SDS mitigation and adaptation. Moreover, population increases or social pressures such as poverty or the need for immigration in search of food and security have been shown to affect lands and promote desertification. The ageing population will also be a challenge for competent authorities because they will need to increase their organisation, infrastructure, and protection mechanisms to cope with the potential increase in SDS due to climate change or humanitarian crises. Lastly, shortage of basic resources such as water can result to conflicts between countries and internal instability leading to the abandonment of land and SLWM which can promote SDS, thus affecting areas far away from sources. Thus, tools such as the Sendai Framework for Disaster Risk Reduction are best to be used before adopting measures.

For further research, the socioeconomic and political teleconnections (i.e. an action in an area affecting another area far away) between SDS sources and the EMR can be studied because this connection can result in major SDS. Such a teleconnection can be because the source areas affecting the EMR have been troubled by non-peaceful conflicts, as in the case of Syria, and this can further degrade the lands and drive farmers away, enhancing dust suspension or forming new dust sources.

In closing, the EMR is an area affected by different SDS source areas, each of different social, political, and economic dimensions, necessitating a customised approach for implementing wind erosion mitigation measures. Notably, adaptation measures are mainly based on early warning systems and progress accordingly with capacity building to provide accurate forecasts. For the proper use of accurate forecasts, more infrastructure, organisation, and will from competent authorities and employers to implement adaptation measures are required. The socioeconomic and political context for implementing various mitigation and adaptation measures should always be examined before proceeding. This review can guide policymakers, planners, and engineers to what they should consider before selecting and implementing measures to mitigate or adapt to SDS.

Funding Open access funding provided by the Cyprus Libraries Consortium (CLC). This work was supported by the Research and Innovation Foundation (RIF) of Cyprus within the framework of the Isles of Excellence Program under Grant Agreement No. EXCELLENCE/1216/0294. Moreover, A. Eleftheriou, M. Neophytou, and P. Yiallouros are grateful for the financial support from the LIFE+-MEDEA Program under Grant Agreement LIFE16 CCA/CY/000041. Prashant Kumar acknowledges the support received through the OPERANDUM (OPEn-air laboRAtories for Nature baseD solUtions to Manage hydro-meteo risks) project, which is funded by the European Union's Horizon 2020 research and innovation programme under the Grant Agreement No: 776848.

Data availability Data sharing not applicable to this article as no datasets were generated or analysed during the current study.

Declarations

Ethics approval and consent to participate Not applicable.

Consent for publication I would like to declare on behalf of my co-authors that the manuscript has been approved by all authors for publication, and the work described is original research that has not been published previously, and is not under consideration for publication elsewhere, in whole or in part. All of the listed authors have approved the enclosed manuscript.

Conflict of interest The authors declare no competing interests.

Open Access This article is licensed under a Creative Commons Attribution 4.0 International License, which permits use, sharing, adaptation, distribution and reproduction in any medium or format, as long as you give appropriate credit to the original author(s) and the source, provide a link to the Creative Commons licence, and indicate if changes were made. The images or other third party material in this article are included in the article's Creative Commons licence, unless indicated otherwise in a credit line to the material. If material is not included in the article's Creative Commons licence and your intended use is not permitted by statutory regulation or exceeds the permitted use, you will need to obtain permission directly from the copyright holder. To view a copy of this licence, visit http://creativecommons.org/licenses/by/4.0/.

References

- Abdou I (2013) Desertification control in Niger: the medium term action plan 2006–2011. In: Heshmati GA, Squires VR (eds) Combating desertification in Asia, Africa and the Middle East. Springer, Netherlands, Dordrecht, pp 191–213. https://doi.org/10.1007/978-94-007-6652-5_10
- Abdourhamane Toure A, Rajot JL, Garba Z, Marticorena B, Petit C, Sebag D (2011) Impact of very low crop residues cover on wind erosion in the Sahel. CATENA 85:205–214. https://doi.org/10.1016/j. catena.2011.01.002
- Abhijith KV, Gokhale S (2015) Passive control potentials of trees and on-street parked cars in reduction of air pollution exposure in urban street canyons. Environ Pollut 204:99–108. https://doi.org/10. 1016/j.envpol.2015.04.013
- Abhijith KV, Kumar P (2020) Quantifying particulate matter reduction and their deposition on the leaves of green infrastructure. Environ Pollut 265:114884. https://doi.org/10.1016/j.envpol.2020.114884
- Abhijith KV, Kumar P, Gallagher J, McNabola A, Baldauf R, Pilla F, Broderick B, Di Sabatino S, Pulvirenti B (2017) Air pollution abatement performances of green infrastructure in open road and built-up street canyon environments – a review. Atmos Environ 162:71–86. https://doi.org/10. 1016/j.atmosenv.2017.05.014
- Abu-Zreig MM, Tamimi A, Alazba AA (2011) Soil erosion control and moisture conservation of arid lands with stone cover. Arid Land Res Manag 25:294–307. https://doi.org/10.1080/15324982. 2011.565859
- Achilleos S, Evans JS, Yiallouros PK, Kleanthous S, Schwartz J, Koutrakis P (2014) PM 10 concentration levels at an urban and background site in Cyprus: the impact of urban sources and dust storms. J Air Waste Manag Assoc 64:1352–1360. https://doi.org/10.1080/10962247.2014.923061
- Achilleos S, Mouzourides P, Kalivitis N, Katra I, Kloog I, Kouis P, Middleton N, Mihalopoulos N, Neophytou M, Panayiotou A, Papatheodorou S, Savvides C, Tymvios F, Vasiliadou E, Yiallouros P, Koutrakis P (2020) Spatio-temporal variability of desert dust storms in Eastern Mediterranean (Crete, Cyprus, Israel) between 2006 and 2017 using a uniform methodology. Sci Total Environ 714:136693. https://doi.org/10.1016/j.scitotenv.2020.136693
- Achilleos S, Michanikou A, Kouis P, Papatheodorou SI, Panayiotou AG, Kinni P, Mihalopoulos N, Kalivitis N, Kouvarakis G, Galanakis E, Michailidi E, Tymvios F, Chrysanthou A, Neophytou M, Mouzourides P, Savvides C, Vasiliadou E, Papasavvas I, Christophides T, Nicolaou R, Avraamides P, Kang CM, Middleton N, Koutrakis P, Yiallouros PK (2023) Improved indoor air quality during desert dust storms: the impact of the MEDEA exposure-reduction strategies. Sci Total Environ 863:160973. https://doi.org/10.1016/j.scitotenv.2022.160973
- ADF (African Development Fund) (2020) Lake Chad basin sustainable development programme (PRODE-BALT). https://projectsportal.afdb.org/dataportal/VProject/show/P-Z1-CZ0-002. accessed in 2020
- AghaKouchak A, Norouzi H, Madani K, Mirchi A, Azarderakhsh M, Nazemi A, Nasrollahi N, Farahmand A, Mehran A, Hasanzadeh E (2015) Aral Sea syndrome desiccates Lake Urmia: call for action. J Great Lakes Res 41:307–311. https://doi.org/10.1016/j.jglr.2014.12.007
- Al-Ansari N (2016) Hydro-politics of the Tigris and Euphrates basins. Engineering 8(3):140–172. https://doi.org/10.4236/eng.2016.83015
- Al-Hemoud A, Al-Dousari A, Misak R, Al-Sudairawi M, Naseeb A, Al-Dashti H, Al-Dousari N (2019) Economic impact and risk assessment of sand and dust Storms (SDS) on the oil and gas industry in Kuwait. Sustainability 11(1):200. https://doi.org/10.3390/su11010200
- Allafi KAM (2014) The impact of changing agricultural policies on Libyan agricultural performance. Ph. D. Thesis, Sheffield Hallam University, Sheffield, United Kingdom. p 283
- Alloway BJ, Ayres DC, Ayres DC (1997) Chemical principles of environmental pollution, 2nd edn. Blackie Academic & Professional, London
- Armbrust DV, Retta A (2000) Wind and sandblast damage to growing vegetation. Ann Arid Zone 39(3):273–284
- Aslanoğlu SY, Proestakis E, Gkikas A, Güllü G, Amiridis V (2022) Dust climatology of Turkey as a part of the Eastern Mediterranean Basin via 9-year CALIPSO-derived product. Atmosphere 13:733. https://doi.org/10.3390/atmos13050733
- Ates S, Clifton K, Werner J, Louhaichi M (2016) The role of governance in sustainable rangeland management. In: Kyriazopoulos AP, López-Francos A, Porqueddu C, Sklavou P (eds) Ecosystem services and socio-economic benefits of Mediterranean grasslands. CIHEAM, Zaragoza, pp 433–436
- Babatola SS (2018) Global burden of diseases attributable to air pollution. J Public Health Afr 9. https:// doi.org/10.4081/jphia.2018.813

- Badescu V, Cathcart RB (2010) Dune sand fixation: Mauritania seawater pipeline macroproject. In: Badescu V, Cathcart RB (eds) Macro-engineering seawater in unique environments, environmental science and engineering. Springer Berlin Heidelberg, Berlin, Heidelberg, pp 465–488
- Baek S-O, Kim Y-S, Perry R (1997) Indoor air quality in homes, offices and restaurants in Korean urban areas—indoor/outdoor relationships. Atmos Environ 31:529–544. https://doi.org/10.1016/S1352-2310(96)00215-4
- Barwise Y, Kumar P (2020) Designing vegetation barriers for urban air pollution abatement: a practical review for appropriate plant species selection. npj Clim Atmos Sci 3:12. https://doi.org/10.1038/ s41612-020-0115-3
- Bastin J-F, Finegold Y, Garcia C, Mollicone D, Rezende M, Routh D, Zohner CM, Crowther TW (2019) The global tree restoration potential. Science 365(6448):76–79. https://doi.org/10.1126/science.aax08 48
- Ben Moussa-Machraoui S, Errouissi F, Ben-Hammouda M, Nouira S (2010) Comparative effects of conventional and no-tillage management on some soil properties under Mediterranean semi-arid conditions in northwestern Tunisia. Soil Tillage Res 106:247–253. https://doi.org/10.1016/j.still.2009.10.009
- Ben-Ami Y, Koren I, Rudich Y, Artaxo P, Martin ST, Andreae MO (2010) Transport of North African dust from the Bodélé depression to the Amazon Basin: a case study. Atmos Chem Phys 10(16):7533– 7544. https://doi.org/10.5194/acp-10-7533-2010
- Benedetti A, Reid JS, Knippertz P, Marsham JH, Di Giuseppe F, Rémy S, Basart S, Boucher O, Brooks IM, Menut L, Mona L, Laj P, Pappalardo G, Wiedensohler A, Baklanov A, Brooks M, Colarco PR, Cuevas E, da Silva A, Escribano J, Flemming J, Huneeus N, Jorba O, Kazadzis S, Kinne S, Popp T, Quinn PK, Sekiyama TT, Tanaka T, Terradellas E (2018) Status and future of numerical atmospheric aerosol prediction with a focus on data requirements. Atmos Chem Phys 18:10615–10643. https://doi.org/10.5194/acp-18-10615-2018
- Benjaminsen TA, Hiernaux P (2019) From desiccation to global climate change: a history of the desertification narrative in the West African Sahel, 1900–2018. Global Environment 12(1):206–236. https://doi. org/10.3197/ge.2019.120109
- Betzer PR, Carder KL, Duce RA, Merrill JT, Tindale NW, Uematsu M, Costello DK, Young RW, Feely RA, Breland JA, Bernstein RE, Greco AM (1988) Long–range transport of giant mineral aerosol particles. Nature 336:568–571. https://doi.org/10.1038/336568a0
- Beyene F (2010) Locating the adverse effects of rangeland enclosure among herders in eastern Ethiopia. Land Use Policy 27:480–488. https://doi.org/10.1016/j.landusepol.2009.07.001
- Bichet A, Diedhiou A (2018) West African Sahel has become wetter during the last 30 years, but dry spells are shorter and more frequent. Clim Res 75:155–162. https://doi.org/10.3354/cr01515
- Bielders CL, John PAL, Karlheinz M (2001) Wind erosion control technologies in the West African Sahel: the effectiveness of windbreaks, mulching and soil tillage, and the perspective of farmers. Ann Arid Zone 40(3):369–394
- Bodenheimer S, Lensky IM, Dayan U (2019) Characterization of Eastern Mediterranean dust storms by area of origin; North Africa vs. Arabian Peninsula. Atmos Environ 198:158–165. https://doi.org/10.1016/j. atmosenv.2018.10.034
- Bot A, Benites J (2005) The importance of soil organic matter: key to drought-resistant soil and sustained food production, FAO soils bulletin. Food and Agriculture Organization of the United Nations, Rome
- Bouwer LM, Bubeck P, Aerts JCJH (2010) Changes in future flood risk due to climate and development in a Dutch polder area. Glob Environ Chang 20:463–471. https://doi.org/10.1016/j.gloenvcha.2010.04.002
- Butt B (2016) Ecology, mobility and labour: dynamic pastoral herd management in an uncertain world: -EN- -FR- Écologie, mobilité et travail : dynamique pastorale de la gestion des troupeaux dans un monde incertain -ES- Ecología, movilidad y trabajo. Gestión dinámica del rebaño ganadero en un mundo incierto. Rev Sci Tech OIE 35:461–472. https://doi.org/10.20506/rst.35.2.2530
- Çapraz Ö, Deniz A (2021) Particulate matter (PM₁₀ and PM_{2.5}) concentrations during a Saharan dust episode in Istanbul. Air Qual Atmos Health 14:109–116. https://doi.org/10.1007/s11869-020-00917-4
- Chen R, Lee I, Zhang L (2015) Biopolymer stabilization of mine tailings for dust control. J Geotech Geoenviron Eng 141:04014100. https://doi.org/10.1061/(ASCE)GT.1943-5606.0001240
- Chen K, Vicedo-Cabrera AM, Dubrow R (2020) Projections of ambient temperature- and air pollutionrelated mortality burden under combined climate change and population aging scenarios: a review. Curr Envir Health Rpt 7:243–255. https://doi.org/10.1007/s40572-020-00281-6
- Choobari OA, Zawar-Reza P, Sturman A (2014) The global distribution of mineral dust and its impacts on the climate system: a review. Atmos Res 138:152–165. https://doi.org/10.1016/j.atmosres.2013. 11.007

- Chow TL, Rees HW, Monteith JO, Toner P, Lavoie J (2007) Effects of coarse fragment content on soil physical properties, soil erosion and potato production. Can J Soil Sci 87:565–577. https://doi.org/10. 4141/CJSS07006
- Cohen AJ, Brauer M, Burnett R, Anderson HR, Frostad J, Estep K, Balakrishnan K, Brunekreef B, Dandona L, Dandona R, Feigin V, Freedman G, Hubbell B, Jobling A, Kan H, Knibbs L, Liu Y, Martin R, Morawska L, Pope CA, Shin H, Straif K, Shaddick G, Thomas M, van Dingenen R, van Donkelaar A, Vos T, Murray CJL, Forouzanfar MH (2017) Estimates and 25-year trends of the global burden of disease attributable to ambient air pollution: an analysis of data from the Global Burden of Diseases Study 2015. Lancet 389:1907–1918. https://doi.org/10.1016/S0140-6736(17)30505-6
- Conceição R, Silva HG, Mirão J, Gostein M, Fialho L, Narvarte L, Collares-Pereira M (2018) Saharan dust transport to Europe and its impact on photovoltaic performance: a case study of soiling in Portugal. Sol Energy 160:94–102. https://doi.org/10.1016/j.solener.2017.11.059
- Coppock DL, Fernández-Giménez M, Hiernaux P, Huber-Sannwald E, Schloeder C, Valdivia C, Arredondo JT, Jacobs M, Turin C, Turner M (2017) Rangeland systems in developing nations: conceptual advances and societal implications. In: Briske DD (ed) Rangeland systems. Springer International Publishing, Cham, pp 569–641. https://doi.org/10.1007/978-3-319-46709-2_17
- Csavina J, Landázuri A, Wonaschütz A, Rine K, Rheinheimer P, Barbaris B, Conant W, Sáez AE, Betterton EA (2011) Metal and metalloid contaminants in atmospheric aerosols from mining operations. Water Air Soil Pollut 221:145–157. https://doi.org/10.1007/s11270-011-0777-x
- Csavina J, Field J, Taylor MP, Gao S, Landázuri A, Betterton EA, Sáez AE (2012) A review on the importance of metals and metalloids in atmospheric dust and aerosol from mining operations. Sci Total Environ 433:58–73. https://doi.org/10.1016/j.scitotenv.2012.06.013
- Cuesta J, Marsham JH, Parker DJ, Flamant C (2009) Dynamical mechanisms controlling the vertical redistribution of dust and the thermodynamic structure of the West Saharan atmospheric boundary layer during summer. Atmosph Sci Lett 10:34–42. https://doi.org/10.1002/asl.207
- Danesh-Yazdi M, Ataie-Ashtiani B (2019) Lake Urmia crisis and restoration plan: planning without appropriate data and model is gambling. J Hydrol 576:639–651. https://doi.org/10.1016/j.jhydrol. 2019.06.068
- Daoui K, Fatemi ZEA (2014) Agroforestry systems in Morocco: the case of olive tree and annual crops association in Saïs Region. In: Behnassi M, Shahid SA, Mintz-Habib N (eds) Science, Policy and Politics of Modern Agricultural System. Springer Netherlands, Dordrecht, pp 281–289. https://doi. org/10.1007/978-94-007-7957-0_19+
- Davies J, Poulsen L, Schulte-Herbrüggen B, MacKinnon K, Crawhall N, Henwood WD, Dudley N, Smith J, Gudka M (2012) Conserving dryland biodiversity. International Union for Conservation of Nature and Natural Resources, Nairobi, Kenya
- de la Paz D, Vedrenne M, Borge R, Lumbreras J, Manuel de Andrés J, Pérez J, Rodríguez E, Karanasiou A, Moreno T, Boldo E, Linares C (2013) Modelling Saharan dust transport into the Mediterranean basin with CMAQ. Atmos Environ 70:337–350. https://doi.org/10.1016/j.atmosenv.2013.01.013
- Dong S, Wolf SA, Lassoie JP, Liu S, Long R, Yi S, Jasra AW, Phuntsho K (2017) Bridging the gaps between science and policy for the sustainable management of rangeland resources in the developing world. Bioscience 67:656–663. https://doi.org/10.1093/biosci/bix042
- Doyle JD, Durran DR (2002) The dynamics of mountain-wave-induced rotors. J Atmos Sci 59:186–201. https://doi.org/10.1175/1520-0469(2002)059%3c0186:TDOMWI%3e2.0.CO;2
- Dregne HE (2002) Land degradation in the drylands. Arid Land Res Manag 16:99–132. https://doi.org/ 10.1080/153249802317304422
- Eleftheriou A, Mouzourides P, Neophytou MK-A (2021) How accurate are dust surface concentrations forecasts from numerical models? A preliminary analysis of the multi-model median forecasting in Eastern Mediterranean Region. In: Mensink C, Matthias V (eds) Air Pollution Modeling and Its Application XXVII, Springer Proceedings in Complexity. Springer Berlin Heidelberg, Berlin, Heidelberg, pp 337–344. https://doi.org/10.1007/978-3-662-63760-9_49
- Evan AT, Flamant C, Fiedler S, Doherty O (2014) An analysis of aeolian dust in climate models: AEO-LIAN DUST IN CMIP5. Geophys Res Lett 41:5996–6001. https://doi.org/10.1002/2014GL0605 45
- FAO (Food and Agriculture Organization of the United Nation), ITPS (Intergovernmental Technical Panel on Soils) (2015) Status of the World's Soil Resources, Rome, p 648
- Faysse N (2015) The rationale of the Green Morocco Plan: missing links between goals and implementation. J North Afr Stud 20:622–634. https://doi.org/10.1080/13629387.2015.1053112
- Feitelson E, Tubi A (2017) A main driver or an intermediate variable? Climate change, water and security in the Middle East. Glob Environ Chang 44:39–48. https://doi.org/10.1016/j.gloenvcha.2017. 03.001

- Fensholt R, Mbow C, Brandt M, Rasmussen K (2017) Desertification and re-greening of the Sahel. Oxford Univ Press. https://doi.org/10.1093/acrefore/9780190228620.013.553
- Fernández AJ, Sicard M, Costa MJ, Guerrero-Rascado JL, Gómez-Amo JL, Molero F, Barragán R, Basart S, Bortoli D, Bedoya-Velásquez AE, Utrillas MP, Salvador P, Granados-Muñoz MJ, Potes M, Ortiz-Amezcua P, Martínez-Lozano JA, Artíñano B, Muñoz-Porcar C, Salgado R, Román R, Rocadenbosch F, Salgueiro V, Benavent-Oltra JA, Rodríguez-Gómez A, Alados-Arboledas L, Comerón A, Pujadas M (2019) Extreme, wintertime Saharan dust intrusion in the Iberian Peninsula: lidar monitoring and evaluation of dust forecast models during the February 2017 event. Atmos Res 228:223–241. https://doi.org/10.1016/j.atmosres.2019.06.007
- Forzieri G, Feyen L, Russo S, Vousdoukas M, Alfieri L, Outten S, Migliavacca M, Bianchi A, Rojas R, Cid A (2016) Multi-hazard assessment in Europe under climate change. Clim Change 137:105– 119. https://doi.org/10.1007/s10584-016-1661-x
- Fryrear DW (1984) Soil ridges-clods and wind erosion. Trans ASAE 27:445–448. https://doi.org/10.13031/ 2013.32808
- Galeazzi G, Medinilla A, Ebiede TM, Desmidt S (2017) Understanding the Lake Chad Basin Commission (LCBC). https://ecdpm.org/wp-content/uploads/LCBC-Background-Paper-PEDRO-Political-Econo my-Dynamics-Regional-Organisations-Africa-ECDPM-2017.pdf. accessed in 2020
- Gallisai R, Peters F, Volpe G, Basart S, Baldasano JM (2014) Saharan dust deposition may affect phytoplankton growth in the Mediterranean Sea at ecological time scales. PLoS One 9(10):e110762. https://doi.org/10.1371/journal.pone.0110762
- Ganjalinejhad M et al. (2018) A review on national and international legal documents on combating sand and dust storms. Anthropogenic Pollut J 2(1). https://doi.org/10.22034/ap.2018.538375
- Ganor E, Osetinsky I, Stupp A, Alpert P (2010) Increasing trend of African dust, over 49 years, in the eastern Mediterranean. J Geophys Res 115:D07201. https://doi.org/10.1029/2009JD012500
- Garousi V, Najafi A, Samadi A, Rasouli K, Khanaliloo B (2013) Environmental crisis in Lake Urmia, Iran: a systematic review of causes, negative consequences and possible solutions. Proceedings of the 6th International Perspective on Water Resources & the Environment (IPWE) Izmir, Turkey
- Garrison VH, Majewski MS, Konde L, Wolf RE, Otto RD, Tsuneoka Y (2014) Inhalable desert dust, urban emissions, and potentially biotoxic metals in urban Saharan-Sahelian air. Sci Total Environ 500– 501:383–394. https://doi.org/10.1016/j.scitotenv.2014.08.106
- Gebrehiwot T, Veen AVD (2014) The effect of enclosure in rehabilitating degraded vegetation: a case of Enderta District. Northern Ethiopia for Res 3:128. https://www.longdom.org/open-access-pdfs/theeffect-ofenclosures-in-rehabilitating-degraded-vegetation-2168-9776.1000128.pdf
- Geist HJ, Lambin EF (2004) Dynamic causal patterns of desertification. Bioscience 54:817. https://doi.org/ 10.1641/0006-3568(2004)054[0817:DCPOD]2.0.CO;2
- Gherboudj I, Naseema Beegum S, Ghedira H (2017) Identifying natural dust source regions over the Middle-East and North-Africa: estimation of dust emission potential. Earth Sci Rev 165:342–355. https:// doi.org/10.1016/j.earscirev.2016.12.010
- Ghose MK (2002) Air pollution due to opencast coal mining and the characteristics of air-borne dust–an Indian scenario. Int J Environ Stud 59:211–228. https://doi.org/10.1080/00207230210927
- Gill TE (1996) Eolian sediments generated by anthropogenic disturbance of playas: human impacts on the geomorphic system and geomorphic impacts on the human system. Geomorphology 17:207–228. https://doi.org/10.1016/0169-555X(95)00104-D
- Gil-Loaiza J, Field JP, White SA, Csavina J, Felix O, Betterton EA, Sáez AE, Maier RM (2018) Phytoremediation reduces dust emissions from metal(loid)-contaminated Mine tailings. Environ Sci Technol 52:5851–5858. https://doi.org/10.1021/acs.est.7b05730
- Ginoux P (2017) Warming or cooling dust? Nat Geosci 10(4):246–248. https://doi.org/10.1038/ngeo2923
- Ginoux P, Prospero J, Torres O, Chin M (2004) Long-term simulation of global dust distribution with the GOCART model: correlation with North Atlantic Oscillation. Environ Model Softw 19:113–128. https://doi.org/10.1016/S1364-8152(03)00114-2
- Ginoux P, Prospero JM, Gill TE, Hsu NC, Zhao M (2012) Global-scale attribution of anthropogenic and natural dust sources and their emission rates based on MODIS Deep Blue aerosol products: anthropogenic and Natural Dust Sources. Rev Geophys 50(3). https://doi.org/10.1029/2012RG000388
- GIZ (2012) Good practices in soil and water conservation, https://www.echocommunity.org/en/resources/ c411bec7-7ae5-4b89-b5b2-d790017e6fe9. accessed in 2020
- Goudie AS, Middleton NJ (1992) The changing frequency of dust storms through time. Clim Change 20:197–225. https://doi.org/10.1007/BF00139839
- Goudie A (2018) Dust storms and ephemeral lakes. Desert 23. https://doi.org/10.22059/jdesert.2018.66370
- Govers G, Merckx R, van Wesemael B, Van Oost K (2017) Soil conservation in the 21st century: why we need smart agricultural intensification. SOIL 3(1):45–59. https://doi.org/10.5194/soil-3-45-2017

- Griffin DW, Kellogg CA, Shinn EA (2001) Dust in the wind: long range transport of dust in the atmosphere and its implications for global public and ecosystem health. Glob Change Human Health 2:20–33. https://doi.org/10.1023/A:1011910224374
- Han G, Wu Q, Tang Y (2011) Acid rain and alkalization in southwestern China: chemical and strontium isotope evidence in rainwater from Guiyang. J Atmos Chem 68(2):139–155. https://doi.org/10.1007/ s10874-012-9213-x
- Heal MR, Kumar P, Harrison RM (2012) Particles, air quality, policy and health. Chem Soc Rev 41(19):6606. https://doi.org/10.1039/c2cs35076a
- Hoek G (2017) Methods for assessing long-term exposures to outdoor air pollutants. Curr Envir Health Rpt 4:450–462. https://doi.org/10.1007/s40572-017-0169-5
- Holben BN, Eck TF, Slutsker I, Tanré D, Buis JP, Setzer A, Vermote E, Reagan JA, Kaufman YJ, Nakajima T, Lavenu F, Jankowiak I, Smirnov A (1998) AERONET—a federated instrument network and data archive for aerosol characterization. Remote Sens Environ 66:1–16. https://doi.org/10.1016/S0034-4257(98)00031-5
- Homrani Bakali A, Acherckouf M, Maatougui A, Mrabet R, Slimani M (2016) Rangeland rehabilitation using rainwater harvesting and rosemary (Rosmarinus officinalis L.) transplantation in the Southeast of Morocco. In: Kyriazopoulos AP, López-Francos A, Porqueddu C, Sklavou P (eds) Ecosystem services and socio-economic benefits of Mediterranean grasslands, Zaragoza, pp 395–398
- Hong C, Zhang Q, Zhang Y, Davis SJ, Tong D, Zheng Y, Liu Z, Guan D, He K, Schellnhuber HJ (2019) Impacts of climate change on future air quality and human health in China. Proc Natl Acad Sci USA 116:17193–17200. https://doi.org/10.1073/pnas.1812881116
- Houyou Z, Bielders CL, Benhorma HA, Dellal A, Boutemdjet A (2016) Evidence of strong land degradation by wind erosion as a result of rainfed cropping in the Algerian steppe: a case study at Laghouat. Land Degrad Develop 27:1788–1796. https://doi.org/10.1002/ldr.2295
- Huertas JI, Huertas ME, Cervantes G, Díaz J (2014) Assessment of the natural sources of particulate matter on the opencast mines air quality. Sci Total Environ 493:1047–1055. https://doi.org/10.1016/j.scito tenv.2014.05.111
- Hui WJ, Cook BI, Ravi S, Fuentes JD, D'Odorico P (2008) Dust-rainfall feedbacks in the West African Sahel: DUST-RAINFALL FEEDBACKS. Water Resour Research 44(5). https://doi.org/10.1029/ 2008WR006885
- Huszar PC, Piper SL (1986) Estimating the off-site costs of wind erosion in New Mexico. J Soil Water Conserv 41(6):414–416
- Ikazaki K, Shinjo H, Tanaka U, Tobita S, Funakawa S, Kosaki T (2011) "Fallow Band System", a land management practice for controlling desertification and improving crop production in the Sahel, West Africa. 1. Effectiveness in desertification control and soil fertility improvement. Soil Sci Plant Nutr 57:573–586. https://doi.org/10.1080/00380768.2011.593155
- Im H-J, Kwon H-J, Ha M, Lee SG, Hwang S-S, Ha EH, Cho S-H (2006) Public perceptions of the risk of Asian dust storms in Seoul and its metropolitan area. J Prev Med Public Health 39:205–212
- IUCN (International Union for Conservation of Nature) (2017) Drylands and land degradation. https://www. iucn.org/resources/issues-brief/drylands-and-land-degradation
- Jaafar HH, Woertz E (2016) Agriculture as a funding source of ISIS: a GIS and remote sensing analysis. Food Policy 64:14–25. https://doi.org/10.1016/j.foodpol.2016.09.002
- Jackson JK (1984) Why do forest plantation fail. Proceedings of the 100 years of forestly education and research in the Netherlands, September 19–23, 1983, Wageningen, Netherlands, pp 277–285
- Janhäll S (2015) Review on urban vegetation and particle air pollution deposition and dispersion. Atmos Environ 105:130–137. https://doi.org/10.1016/j.atmosenv.2015.01.052
- Joseph PV, Raipal DK, Deka SN (1980) "Andhi", the convective duststorm of northwest India. MAUSAM 31:431-442. https://doi.org/10.54302/mausam.v31i3.3781
- Kabisch N, Frantzeskaki N, Pauleit S, Naumann S, Davis M, Artmann M, Haase D, Knapp S, Korn H, Stadler J, Zaunberger K, Bonn A (2016) Nature-based solutions to climate change mitigation and adaptation in urban areas: perspectives on indicators, knowledge gaps, barriers, and opportunities for action. E&S 21:39. https://doi.org/10.5751/ES-08373-210239
- Karra S, Malki-Epshtein L, Neophytou MK-A (2017) Air flow and pollution in a real, heterogeneous urban street canyon: a field and laboratory study. Atmos Environ 165:370–384. https://doi.org/10.1016/j. atmosenv.2017.06.035
- Kaskaoutis DG, Kambezidis HD, Nastos PT, Kosmopoulos PG (2008) Study on an intense dust storm over Greece. Atmos Environ 42:6884–6896. https://doi.org/10.1016/j.atmosenv.2008.05.017
- Katra I, Lavee H, Sarah P (2008) The effect of rock fragment size and position on topsoil moisture on arid and semi-arid hillslopes. CATENA 72:49–55. https://doi.org/10.1016/j.catena.2007.04.001

- Katra I, Laor S, Swet N, Kushmaro A, Ben-Dov E (2017) Shifting cyanobacterial diversity in response to agricultural soils associated with dust emission. Land Degrad Develop 28:878–886. https://doi.org/ 10.1002/ldr.2644
- Kaufman YJ, Koren I, Remer LA, Rosenfeld D, Rudich Y (2005) The effect of smoke, dust, and pollution aerosol on shallow cloud development over the Atlantic Ocean. Proc Natl Acad Sci 102(32):11207– 11212. https://doi.org/10.1073/pnas.0505191102
- Kim D, Chin M, Bian H, Tan Q, Brown ME, Zheng T, You R, Diehl T, Ginoux P, Kucsera T (2013) The effect of the dynamic surface bareness on dust source function, emission, and distribution: DYNAMIC DUST SOURCE FUNCTION. J Geophys Res Atmos 118:871–886. https://doi.org/10. 1029/2012JD017907
- Kinni P, Kouis P, Dimitriou H, Yarza S, Papatheodorou SI, Kampriani E, Charalambous M, Middleton N, Novack V, Galanakis E, Yiallouros PK (2021) Health effects of desert dust storm events in the south-eastern Mediterranean: perceptions and practices of local stakeholders. East Mediterr Health J 27:1092–1101. https://doi.org/10.26719/emhj.21.037
- Knippertz P, Stuut J-BW (eds) (2014) Mineral dust: A key player in the earthsystem. Springer Netherlands, Dordrecht. https://doi.org/10.1007/978-94-017-8978-3
- Kok JF, Ridley DA, Zhou Q, Miller RL, Zhao C, Heald CL, Ward DS, Albani S, Haustein K (2017) Smaller desert dust cooling effect estimated from analysis of dust size and abundance. Nat Geosci 10(4):274– 278. https://doi.org/10.1038/ngeo2912
- Koren I, Kaufman YJ, Washington R, Todd MC, Rudich Y, Martins JV, Rosenfeld D (2006) The Bodélé depression: a single spot in the Sahara that provides most of the mineral dust to the Amazon forest. Environ Res Lett 1(1):014005. https://doi.org/10.1088/1748-9326/1/1/014005
- Kouis P, Papatheodorou SI, Kakkoura MG, Middleton N, Galanakis E, Michaelidi E, Achilleos S, Mihalopoulos N, Neophytou M, Stamatelatos G, Kaniklides C, Revvas E, Tymvios F, Savvides C, Koutrakis P, Yiallouros PK (2021) The MEDEA childhood asthma study design for mitigation of desert dust health effects: implementation of novel methods for assessment of air pollution exposure and lessons learned. BMC Pediatr 21:13. https://doi.org/10.1186/s12887-020-02472-4
- Kouis P, Michanikou A, Galanakis E, Michaelidou E, Dimitriou H, Perez J, Kinni P, Achilleos S, Revvas E, Stamatelatos G, Zacharatos H, Savvides C, Vasiliadou E, Kalivitis N, Chrysanthou A, Tymvios F, Papatheodorou SI, Koutrakis P, Yiallouros PK (2023) Responses of schoolchildren with asthma to recommendations to reduce desert dust exposure: results from the LIFE-MEDEA intervention project using wearable technology. Sci Total Environ 860:160518. https://doi.org/10.1016/j.scitotenv.2022. 160518
- Krasnov H, Katra I, Friger M (2016a) Increase in dust storm related PM10 concentrations: a time series analysis of 2001–2015. Environ Pollut 213:36–42. https://doi.org/10.1016/j.envpol.2015.10.021
- Krasnov H, Kloog I, Friger M, Katra I (2016b) The spatio-temporal distribution of particulate matter during natural dust episodes at an urban scale. PloS One 11(8). https://doi.org/10.1371/journal.pone.0160800
- Kubilay N, Nickovic S, Moulin C, Dulac F (2000) An illustration of the transport and deposition of mineral dust onto the eastern Mediterranean. Atmos Environ 34:1293–1303. https://doi.org/10.1016/S1352-2310(99)00179-X
- Kubilay A, Neophytou MK-A, Matsentides S, Loizou M, Carmeliet J (2017) The pollutant removal capacity of urban street canyons as quantified by the pollutant exchange velocity. Urban Climate 21:136–153. https://doi.org/10.1016/j.uclim.2017.06.003
- Kumar P, Sokolik IN, Nenes A (2011) Measurements of cloud condensation nuclei activity and droplet activation kinetics of fresh unprocessed regional dust samples and minerals. Atmos Chem Phys 11(7):3527–3541. https://doi.org/10.5194/acp-11-3527-2011
- Kumar P, Druckman A, Gallagher J, Gatersleben B, Allison S, Eisenman TS, Hoang U, Hama S, Tiwari A, Sharma A, Abhijith KV, Adlakha D, McNabola A, Astell-Burt T, Feng X, Skeldon AC, de Lusignan S, Morawska L (2019) The nexus between air pollution, green infrastructure and human health. Environ Int 133:105181. https://doi.org/10.1016/j.envint.2019.105181
- Kumar P, Debele SE, Sahani J, Rawat N, Marti-Cardona B, Alfieri SM, Basu B, Basu AS, Bowyer P, Charizopoulos N, Jaakko J (2021) An overview of monitoring methods for assessing the performance of nature-based solutions against natural hazards. Earth Sci Rev 217:103603. https://doi.org/10.1016/j. earscirev.2021.103603
- Kumar P, Zavala-Reyes JC, Tomson M, Kalaiarasan G (2022) Understanding the effects of roadside hedges on the horizontal and vertical distributions of air pollutants in street canyons. Environ Int 158:106883. https://doi.org/10.1016/j.envint.2021.106883
- Labiadh M, Bergametti G, Kardous M, Perrier S, Grand N, Attoui B, Sekrafi S, Marticorena B (2013) Soil erosion by wind over tilled surfaces in South Tunisia. Geoderma 202–203:8–17. https://doi.org/10. 1016/j.geoderma.2013.03.007

- Lancaster N (2011) Desert dune processes and dynamics. In: Thomas DSG (ed) Arid zone geomorphologY. John Wiley & Sons Ltd, Chichester, UK, pp 487–515. https://doi.org/10.1002/9780470710777.ch19
- Larwanou M, Abdoulaye A, Reij C (2006) La régénération Naturelle Assistée dans la région de Zinder (Niger). https://www.formad-environnement.org/RNA_Zinder_usaid.pdf. accessed in 2020
- Lau KM, Kim KM (2007) Cooling of the Atlantic by Saharan dust: cooling of the Atlantic by Saharan dust. Geophys Res Lett 34(23). https://doi.org/10.1029/2007GL031538
- LCBC (Lake Chad Basin Commission) (2016) Lake chad development and climate resilience action plan, http://documents.worldbank.org/curated/en/489801468186879029/pdf/Main-report.pdf,accessed in 2020
- Leenders JK, van Boxel JH, Sterk G (2007) The effect of single vegetation elements on wind speed and sediment transport in the Sahelian zone of Burkina Faso, Earth Surf. Process Landforms 32:1454–1474. https://doi.org/10.1002/esp.1452
- Leenders JK, Sterk G, Van Boxel JH (2011) Modelling wind-blown sediment transport around single vegetation elements: modelling vegetation effects on wind erosion, Earth Surf. Process Landforms 36:1218–1229. https://doi.org/10.1002/esp.2147
- Leenders JK, Sterk G, Boxel JH (2016) Wind erosion reduction by scattered woody vegetation in farmers' fields in Northern Burkina Faso, Land Degrad. Develop 27:1863–1872. https://doi.org/10.1002/ldr. 2322
- Lelieveld J, Evans JS, Fnais M, Giannadaki D, Pozzer A (2015) The contribution of outdoor air pollution sources to premature mortality on a global scale. Nature 525:367–371. https://doi.org/10.1038/natur e15371
- Leung DYC (2015) Outdoor-indoor air pollution in urban environment: challenges and opportunity. Front Environ Sci 2. https://doi.org/10.3389/fenvs.2014.00069
- Levin Z (2005) On the interactions of mineral dust, sea-salt particles, and clouds: a measurement and modeling study from the Mediterranean Israeli Dust Experiment campaign. J Geophys Res 110(D20). https://doi.org/10.1029/2005JD005810
- LIFE+ MEDEA (2020) Mitigating the health effects of desert dust storms using exposure-reduction approaches. https://www.life-medea.eu. accessed in 2020
- Liu D, Abuduwaili J, Wang L (2015) Salt dust storm in the Ebinur Lake region: its 50-year dynamic changes and response to climate changes and human activities. Nat Hazards 77:1069–1080. https://doi.org/10. 1007/s11069-015-1642-9
- Lohmann U, Feichter J (2005) Global indirect aerosol effects: a review. Atmos Chem Phys 5:715–737. https://doi.org/10.5194/acp-5-715-2005
- Mahmood R, Jia S (2019) Assessment of hydro-climatic trends and causes of dramatically declining stream flow to Lake Chad, Africa, using a hydrological approach. Sci Total Environ 675:122–140. https:// doi.org/10.1016/j.scitotenv.2019.04.219
- Mahowald NM, Kloster S, Engelstaedter S, Moore JK, Mukhopadhyay S, McConnell JR, Albani S, Doney SC, Bhattacharya A, Curran MAJ, Flanner MG, Hoffman FM, Lawrence DM, Lindsay K, Mayewski PA, Neff J, Rothenberg D, Thomas E, Thornton PE, Zender CS (2010) Observed 20th century desert dust variability: impact on climate and biogeochemistry. Atmos Chem Phys 10:10875–10893. https://doi.org/10.5194/acp-10-10875-2010
- Mahowald N, Albani S, Kok JF, Engelstaeder S, Scanza R, Ward DS, Flanner MG (2014) The size distribution of desert dust aerosols and its impact on the Earth system. Aeol Res 15:53–71. https://doi.org/10. 1016/j.aeolia.2013.09.002
- Mamouri R-E, Ansmann A, Nisantzi A, Solomos S, Kallos G, Hadjimitsis DG (2016) Extreme dust storm over the eastern Mediterranean in September 2015: satellite, lidar, and surface observations in the Cyprus region. Atmos Chem Phys 16(21):13711–13724. https://doi.org/10.5194/acp-16-13711-2016
- Menut L (2008) Sensitivity of hourly Saharan dust emissions to NCEP and ECMWF modeled wind speed. J Geophys Res 113:D16201. https://doi.org/10.1029/2007JD009522
- Middleton NJ (2017) Desert dust hazards: a global review. Aeol Res 24:53–63. https://doi.org/10.1016/j. aeolia.2016.12.001
- Middleton N (2018) Rangeland management and climate hazards in drylands: dust storms, desertification and the overgrazing debate. Nat Hazards 92:57–70. https://doi.org/10.1007/s11069-016-2592-6
- Middleton N, Kang U (2017) Sand and dust storms: impact mitigation. Sustainability 9:1053. https://doi. org/10.3390/su9061053
- Middleton N, Thomas DSG (1997) World atlas of desertification, 2nd edn. Arnold, New York
- Middleton N, Yiallouros P, Kleanthous S, Kolokotroni O, Schwartz J, Dockery DW, Demokritou P, Koutrakis P (2008) A 10-year time-series analysis of respiratory and cardiovascular morbidity in Nicosia, Cyprus: the effect of short-term changes in air pollution and dust storms. Environ Health 7:39. https:// doi.org/10.1186/1476-069X-7-39

- Middleton N, Stringer L, Goudie A, Thomas D (2011) The forgotten billion: MDG achievement in the drylands. UNDP-UNCCD, New York, p 64
- Miehe S, Kluge J, Von Wehrden H, Retzer V (2010) Long-term degradation of Sahelian rangeland detected by 27 years of field study in Senegal: long-term rangeland monitoring in the Sahel. J Appl Ecol 47:692–700. https://doi.org/10.1111/j.1365-2664.2010.01815.x
- Miller RL, Tegen I, Perlwitz J (2004) Surface radiative forcing by soil dust aerosols and the hydrologic cycle. J Geophys Res 109:n/a-n/a. https://doi.org/10.1029/2003JD004085
- Mokadem N, Hamed Y, Sâad AB, Gargouri I (2014) Atmospheric pollution in North Africa (ecosystems– atmosphere interactions): a case study in the mining basin of El Guettar–M'Dilla (southwestern Tunisia). Arab J Geosci 7:2071–2079. https://doi.org/10.1007/s12517-013-0852-2
- Monjezi M, Shahriar K, Dehghani H, Samimi Namin F (2009) Environmental impact assessment of open pit mining in Iran. Environ Geol 58:205–216. https://doi.org/10.1007/s00254-008-1509-4
- Monteiro A, Basart S, Kazadzis S, Votsis A, Gkikas A, Vandenbussche S, Tobias A, Gama C, García-Pando CP, Terradellas E, Notas G, Middleton N, Kushta J, Amiridis V, Lagouvardos K, Kosmopoulos P, Kotroni V, Kanakidou M, Mihalopoulos N, Kalivitis N, Dagsson-Waldhauserová P, El-Askary H, Sievers K, Giannaros T, Mona L, Hirtl M, Skomorowski P, Virtanen TH, Christoudias T, Di Mauro B, Trippetta S, Kutuzov S, Meinander O, Nickovic S (2022) Multi-sectoral impact assessment of an extreme African dust episode in the Eastern Mediterranean in March 2018. Sci Total Environ 843:156861. https://doi.org/10.1016/j.scitotenv.2022.156861
- Moussadek R, Mrabet R, Dahan R, Zouahri A, El Mourid M, Ranst EV (2014) Tillage system affects soil organic carbon storage and quality in Central Morocco. Appl Environ Soil Sci 2014:1–8. https://doi.org/10.1155/2014/654796
- Mouzourides P, Kumar P, Neophytou MK-A (2015) Assessment of long-term measurements of particulate matter and gaseous pollutants in South-East Mediterranean. Atmos Environ 107:148–165. https://doi.org/10.1016/j.atmosenv.2015.02.031
- Mrabet R (2002) Wheat yield and water use efficiency under contrasting residue and tillage management systems in a semiarid area of morocco. Exp Agric 38(2):237–248. https://doi.org/10.1017/S0014 479702000285
- Mrabet R, Moussadek R, Fadlaoui A, van Ranst E (2012) Conservation agriculture in dry areas of Morocco. Field Crop Res 132:84–94. https://doi.org/10.1016/j.fcr.2011.11.017
- Mulitza S, Heslop D, Pittauerova D, Fischer HW, Meyer I, Stuut J-B, Zabel M, Mollenhauer G, Collins JA, Kuhnert H, Schulz M (2010) Increase in African dust flux at the onset of commercial agriculture in the Sahel region. Nature 466:226–228. https://doi.org/10.1038/nature09213
- Neophytou AM, Yiallouros P, Coull BA, Kleanthous S, Pavlou P, Pashiardis S, Dockery DW, Koutrakis P, Laden F (2013a) Particulate matter concentrations during desert dust outbreaks and daily mortality in Nicosia, Cyprus. J Eposure Sci Environ Epidemiol 23(3):275–280. https://doi.org/10. 1038/jes.2013.10
- Neophytou AM, Yiallouros P, Coull BA, Kleanthous S, Pavlou P, Pashiardis S, Dockery DW, Koutrakis P, Laden F (2013b) Particulate matter concentrations during desert dust outbreaks and daily mortality in Nicosia, Cyprus. J Expo Sci Environ Epidemiol 23:275–280. https://doi.org/10.1038/jes. 2013.10
- Neophytou MK-A, Markides CN, Fokaides PA (2014) An experimental study of the flow through and over two dimensional rectangular roughness elements: deductions for urban boundary layer parameterizations and exchange processes. Phys Fluids 26(8):086603. https://doi.org/10.1063/1.4892979
- Nyssen J, Haile M, Poesen J, Deckers J, Moeyersons J (2006) Removal of rock fragments and its effect on soil loss and crop yield, Tigray, Ethiopia. Soil Use Manag 17:179–187. https://doi.org/10. 1111/j.1475-2743.2001.tb00025.x
- Okpara UT, Stringer LC, Dougill AJ, Bila MD (2015) Conflicts about water in Lake Chad: are environmental, vulnerability and security issues linked? Prog Dev Stud 15:308–325. https://doi.org/10. 1177/1464993415592738
- Ottosen TB, Kumar P (2020) The influence of the vegetation cycle on the mitigation of air pollution by a deciduous roadside hedge. Sustain Cities Soc 53:101919. https://doi.org/10.1016/j.scs.2019. 101919
- Ouria M, Sevinc H (2016) The role of dams in drying up lake Urmia and its environmental impacts on Azerbaijani districts of Iran. Saussurea 6:54–65
- Panagiotou I, Neophytou MK-A, Hamlyn D, Britter RE (2013) City breathability as quantified by the exchange velocity and its spatial variation in real inhomogeneous urban geometries: an example from central London urban area. Sci Total Environ 442:466–477. https://doi.org/10.1016/j.scito tenv.2012.09.001

- Park YM, Kwan M-P (2017) Individual exposure estimates may be erroneous when spatiotemporal variability of air pollution and human mobility are ignored. Health Place 43:85–94. https://doi.org/10. 1016/j.healthplace.2016.10.002
- Park J, Kim K, Lee T, Kim M (2019) Tailings storage facilities (TSFs) dust control using biocompatible polymers. Min Metall Explor 36:785–795. https://doi.org/10.1007/s42461-019-0078-2
- Pausata FSR, Messori G, Zhang Q (2016) Impacts of dust reduction on the northward expansion of the African monsoon during the Green Sahara period. Earth Planet Sci Lett 434:298–307. https://doi. org/10.1016/j.epsl.2015.11.049
- Perini K, Ottelé M, Giulini S, Magliocco A, Roccotiello E (2017) Quantification of fine dust deposition on different plant species in a vertical greening system. Ecol Eng 100:268–276. https://doi.org/10. 1016/j.ecoleng.2016.12.032
- Pey J, Querol X, Alastuey A, Forastiere F, Stafoggia M (2013) African dust outbreaks over the Mediterranean Basin during 2001–2011: PM10 concentrations, phenomenology and trends, and its relation with synoptic and mesoscale meteorology. Atmos Chem Phys 13:1395–1410. https://doi.org/ 10.5194/acp-13-1395-2013
- Pye-Smith C (2013) THE QUIET REVOLUTION: how Niger's farmers are re-greening the parklands of the Sahel. http://apps.worldagroforestry.org/downloads/Publications/PDFS/BL17569.pdf. accessed in 2020
- Rashki A, Kaskaoutis DG, de W Rautenbach CJ, Eriksson PG, Qiang M, Gupta P (2012) Dust storms and their horizontal dust loading in the Sistan region, Iran. Aeolian Res 5:51–62. https://doi.org/10.1016/j. aeolia.2011.12.001
- Reid RS, Fernández-Giménez ME, Galvin KA (2014) Dynamics and resilience of rangelands and pastoral peoples around the globe. Annu Rev Environ Resour 39:217–242. https://doi.org/10.1146/annurevenviron-020713-163329
- Reij C, Tappan G, Smale M (2009) Re-greening the Sahel: farmer-led innovation in Burkina Faso and Niger. https://ebrary.ifpri.org/utils/getfile/collection/p15738coll2/id/130817/filename/131028.pdf. accessed in 2020.
- Reynolds JF, Smith DMS, Lambin EF, Turner BL, Mortimore M, Batterbury SPJ, Downing TE, Dowlatabadi H, Fernandez RJ, Herrick JE, Huber-Sannwald E, Jiang H, Leemans R, Lynam T, Maestre FT, Ayarza M, Walker B (2007) Global desertification: building a science for dryland development. Science 316(5826):847–851. https://doi.org/10.1126/science.1131634
- Riemer N, Ault AP, West M, Craig RL, Curtis JH (2019) Aerosol mixing state: measurements, modeling, and impacts. Rev Geophys 57(2):187–249
- Roberts AJ, Marsham JH, Knippertz P (2015) Disagreements in low-level moisture between (re)analyses over Summertime West Africa. Mon Weather Rev 143:1193–1211. https://doi.org/10.1175/ MWR-D-14-00218.1
- Rowntree K, Duma M, Kakembo V, Thornes J (2004) Debunking the myth of overgrazing and soil erosion. Land Degrad Dev 15(3):203–214
- Sachs JD, Warner AM (1999) The big push, natural resource booms and growth. J Dev Econ 59:43–76. https://doi.org/10.1016/S0304-3878(99)00005-X
- Sæbø A, Popek R, Nawrot B, Hanslin HM, Gawronska H, Gawronski SW (2012) Plant species differences in particulate matter accumulation on leaf surfaces. Sci Total Environ 427–428:347–354. https://doi. org/10.1016/j.scitotenv.2012.03.084
- Safriel U, Adeel Z, Niemeijer D, Puigdefabregas J, White R, Lal R (2005) Dryland systems, in ecosystems and human well-being: current state and trends.: findings of the condition and trends working group. Island Press, pp 623–662
- Saidan M, Albaali AG, Alasis E, Kaldellis JK (2016) Experimental study on the effect of dust deposition on solar photovoltaic panels in desert environment. Renewable Energy 92:499–505. https://doi.org/10. 1016/j.renene.2016.02.031
- Salimi J, Maknoon R, Meijerink S (2019) Designing institutions for watershed management: a case study of the Urmia Lake Restoration National Committee. Water Altern 12(2):609–635
- Samoli E, Nastos PT, Paliatsos AG, Katsouyanni K, Priftis KN (2011) Acute effects of air pollution on pediatric asthma exacerbation: evidence of association and effect modification. Environ Res 111:418–424. https://doi.org/10.1016/j.envres.2011.01.014
- Santoro S, Pluchinotta I, Pagano A, Pengal P, Cokan B, Giordano R (2019) Assessing stakeholders' risk perception to promote nature based solutions as flood protection strategies: the case of the Glinščica river (Slovenia). Sci Total Environ 655:188–201. https://doi.org/10.1016/j.scitotenv.2018.11.116
- SDS-WAS (Sand and Dust Storm Warning Advisory and Assessment System) (2020) Overview & History. https://sds-was.aemet.es/about-us/overview-history, accessed in 2020

- Setton E, Marshall JD, Brauer M, Lundquist KR, Hystad P, Keller P, Cloutier-Fisher D (2011) The impact of daily mobility on exposure to traffic-related air pollution and health effect estimates. J Expo Sci Environ Epidemiol 21:42–48. https://doi.org/10.1038/jes.2010.14
- Shao Y, Dong CH (2006) A review on East Asian dust storm climate, modelling and monitoring. Global Planet Change 52:1–22. https://doi.org/10.1016/j.gloplacha.2006.02.011
- Simoni M, Baldacci S, Maio S, Cerrai S, Sarno G, Viegi G (2015) Adverse effects of outdoor pollution in the elderly. J Thorac Dis 7:34–45. https://doi.org/10.3978/j.issn.2072-1439.2014.12.10
- Soil and Water Conservation Society (U. S.) (2017) Soil erosion research methods, 2nd ed. Routledge. https://doi.org/10.1201/9780203739358
- Solh M, Saxena MC (2011) Food security and climate change in dry areas. Proceedings of an International Conference, February 1–4 2010, Amman, Jordan
- Solheim FS, Vivekanandan J, Ware RH, Rocken C (1999) Propagation delays induced in GPS signals by dry air, water vapor, hydrometeors, and other particulates. J Geophys Res 104(D8):9663–9670. https:// doi.org/10.1029/1999JD900095
- Solmon F, Mallet M, Elguindi N, Giorgi F, Zakey A, Konaré A (2008) Dust aerosol impact on regional precipitation over western Africa, mechanisms and sensitivity to absorption properties. Geophys Res Lett 35:L24705. https://doi.org/10.1029/2008GL035900
- Solomos S, Ansmann A, Mamouri R-E, Binietoglou I, Patlakas P, Marinou E, Amiridis V (2017) Remote sensing and modelling analysis of the extreme dust storm hitting the Middle East and eastern Mediterranean in September 2015. Atmos Chem Phys 17(6):4063–4079. https://doi.org/10.5194/ acp-17-4063-2017
- Solomos S, Abuelgasim A, Spyrou C, Binietoglou I, Nickovic S (2019) Development of a dynamic dust source map for NMME-DREAM v1.0 model based on MODIS Normalized Difference Vegetation Index (NDVI) over the Arabian Peninsula. Geosci Model Dev 12:979–988. https://doi.org/10.5194/ gmd-12-979-2019
- Sommer R, Ryan J, Masri S, Singh M, Diekmann J (2011) Effect of shallow tillage, moldboard plowing, straw management and compost addition on soil organic matter and nitrogen in a dryland barley/ wheat-vetch rotation. Soil Tillage Res 115–116:39–46. https://doi.org/10.1016/j.still.2011.06.003
- Sommer R, Piggin C, Feindel D, Ansar M, van Delden L, Shimonaka K, Abdalla J, Douba O, Estefan G, Haddad A, Haj-Abdo R, Hajdibo A, Hayek P, Khalil Y, Khoder A, Ryan J (2014) Effects of zero tillage and residue retention on soil quality in the Mediterranean Region of Northern Syria. OJSS 04:109–125. https://doi.org/10.4236/ojss.2014.43015
- Sotoudeheian S, Salim R, Arhami M (2016) Impact of Middle Eastern dust sources on PM 10 in Iran: highlighting the impact of Tigris-Euphrates basin sources and Lake Urmia desiccation: Impact of Dust Sources on Iranian Cities. J Geophys Res Atmos 121:14018–14034. https://doi.org/10.1002/2016J D025119
- Stefanski R, Sivakumar MVK (2009) Impacts of sand and dust storms on agriculture and potential agricultural applications of a SDSWS. IOP Conf Ser: Earth Environ Sci 7:012016. https://doi.org/10.1088/ 1755-1307/7/1/012016
- Sterk G (2003) Causes, consequences and control of wind erosion in Sahelian Africa: a review. Land Degrad Dev 14:95–108. https://doi.org/10.1002/ldr.526
- Swap R, Garstang M, Greco S, Talbot R, Kållberg P (1992) Saharan dust in the Amazon Basin. Tellus b: Chem Phys Meteorol 44(2):133–149. https://doi.org/10.3402/tellusb.v44i2.15434
- Tanaka TY, Chiba M (2006) A numerical study of the contributions of dust source regions to the global dust budget. Global Planet Change 52:88–104. https://doi.org/10.1016/j.gloplacha.2006.02.002
- Taylor CM, Belušić D, Guichard F, Parker DJ, Vischel T, Bock O, Harris PP, Janicot S, Klein C, Panthou G (2017) Frequency of extreme Sahelian storms tripled since 1982 in satellite observations. Nature 544:475–478. https://doi.org/10.1038/nature22069
- Thomas DSG, Knight M, Wiggs GFS (2005) Remobilization of southern African desert dune systems by twenty-first century global warming. Nature 435:1218–1221. https://doi.org/10.1038/nature03717
- Tordoff GM, Baker AJM, Willis AJ (2000) Current approaches to the revegetation and reclamation of metalliferous mine wastes. Chemosphere 41:219–228. https://doi.org/10.1016/S0045-6535(99)00414-2
- Triantafyllou E, Diapouli E, Korras-Carraca MB, Manousakas M, Psanis C, Floutsi AA, Spyrou C, Eleftheriadis K, Biskos G (2020) Contribution of locally-produced and transported air pollution to particulate matter in a small insular coastal city. Atmos Pollut Res 11(4):667–678. https://doi.org/10.1016/j. apr.2019.12.015
- Tsiflikiotou MA, Kostenidou E, Papanastasiou DK, Patoulias D, Zarmpas P, Paraskevopoulou D, Diapouli E, Kaltsonoudis C, Florou K, Bougiatioti A, Stavroulas I, Theodosi C, Kouvarakis G, Vasilatou V, Siakavaras D, Biskos G, Pilinis C, Eleftheriadis K, Gerasopoulos E, Mihalopoulos N, Pandis SN

(2019) Summertime particulate matter and its composition in Greece. Atmos Environ 213:597–607. https://doi.org/10.1016/j.atmosenv.2019.06.013

- Tsiouri V, Kakosimos KE, Kumar P (2015) Concentrations, sources and exposure risks associated with particulate matter in the Middle East Area—a review. Air Qual Atmos Health 8:67–80. https://doi.org/ 10.1007/s11869-014-0277-4
- Tsoar H, Zohar Y (1985) Desert dune sand and its potential for modern agricultural development. In: Gradus, Y. (Ed.), Desert development, The GeoJournal Library. Springer Netherlands, Dordrecht, pp 184–200. https://doi.org/10.1007/978-94-009-5396-3_12
- Turner MD, McPeak JG, Ayantunde A (2014) The role of livestock mobility in the livelihood strategies of rural peoples in Semi-Arid West Africa. Hum Ecol 42:231–247. https://doi.org/10.1007/ s10745-013-9636-2
- ULRP (Urmia Lake Restoration Program) (2018) Urmia Lake: lessons and challenges, A Brief Report, p 13 UN (United Nations) (2015) Sendai framework for disaster risk reduction 2015 2030
- UN (United Nations) (2020) World population ageing 2020, New York, p 47
- UNCCD (United Nations Convention to Combat Desertification) (2016) The Great Green Wall, Hope for the Sahara and the Sahel. UNCCD. URL accessed in 2020. https://www.unccd.int/publications/ great-green-wall-hope-sahara-and-sahel
- UNCCD (United Nations Convention to Combat Desertification) (2020) The Great Green Wall, Implementation status and way ahead to 2030
- UNEP (United Nations Environment Programme), WMO (World Meteorological Organization), UNCCD (United Nations Convention to Combat Desertification) (2016) Global Assessment of Sand and Dust Storms, Nairobi, p 139
- UNESCAP (United Nations Economic Commission for Asia and the Pacific) -APDIM (Asian and Pacific Centre for the Development of Disaster Information Management (2020) Guidelines on monitoring and reporting the impacts of sand and dust storms through the Sendai Framework monitoring
- UNESCO (United Nations Educational, Scientific and Cultural Organization) (2004) Managing shared aquifer resources in Africa, Paris, p 238
- van der Does M, Knippertz P, Zschenderlein P, Giles Harrison R, Stuut J-BW (2018) The mysterious long-range transport of giant mineral dust particles. Sci Adv 4:eaau2768. https://doi.org/10.1126/ sciadv.aau2768
- van Ham C, Klimmek H (2017) Partnerships for nature-based solutions in urban areas showcasing successful examples. In: Kabisch N, Korn H, Stadler J, Bonn A (eds) Nature-based solutions to climate change adaptation in urban areas, theory and practice of urban sustainability transitions. Springer International Publishing, Cham, pp 275–289. https://doi.org/10.1007/978-3-319-56091-5_16
- Varga G, Újvári G, Kovács J (2014) Spatiotemporal patterns of Saharan dust outbreaks in the Mediterranean Basin. Aeol Res 15:151–160. https://doi.org/10.1016/j.aeolia.2014.06.005
- Veste M, Littmann T, Breckle S-W, Yair A (2001) The role of biological soil crusts on desert sand dunes in the Northwestern Negev, Israel. In: Breckle S-W, Veste M, Wucherer W (eds) Sustainable Land Use in Deserts. Springer Berlin Heidelberg, Berlin, Heidelberg, pp 357–367. https://doi.org/10. 1007/978-3-642-59560-8_38
- Vodonos A, Friger M, Katra I, Avnon L, Krasnov H, Koutrakis P, Schwartz J, Lior O, Novack V (2014) The impact of desert dust exposures on hospitalizations due to exacerbation of chronic obstructive pulmonary disease. Air Qual Atmos Health 7:433–439. https://doi.org/10.1007/ s11869-014-0253-z
- Vratolis S, Gini MI, Bezantakos S, Stavroulas I, Kalivitis N, Kostenidou E, Louvaris E, Siakavaras D, Biskos G, Mihalopoulos N, Pandis SN, Pilinis C, Papayannis A, Eleftheriadis K (2019) Particle number size distribution statistics at City-Centre Urban Background, urban background, and remote stations in Greece during summer. Atmos Environ 213:711–726. https://doi.org/10.1016/j. atmosenv.2019.05.064
- Vukovic A, Vujadinovic M, Pejanovic G, Andric J, Kumjian MR, Djurdjevic V, Dacic M, Prasad AK, El-Askary HM, Paris BC, Petkovic S, Nickovic S, Sprigg WA (2014) Numerical simulation of "an American haboob." Atmos Chem Phys 14:3211–3230. https://doi.org/10.5194/acp-14-3211-2014
- Wang XM, Zhang CX, Hasi E, Dong ZB (2010) Has the Three Norths Forest Shelterbelt Program solved the desertification and dust storm problems in arid and semiarid China? J Arid Environ 74(1):13– 22. https://doi.org/10.1016/j.jaridenv.2009.08.001
- Washington R, Todd MC, Lizcano G, Tegen I, Flamant C, Koren I, Ginoux P, Engelstaedter S, Bristow CS, Zender CS, Goudie AS, Warren A, Prospero JM (2006) Links between topography, wind, deflation, lakes and dust: the case of the Bodélé Depression. Chad Geophys Res Lett 33:L09401. https://doi.org/10.1029/2006GL025827

- Waylen KA, Hastings EJ, Banks EA, Holstead KL, Irvine RJ, Blackstock KL (2014) The need to disentangle key concepts from ecosystem-approach jargon. Conserv Biol 28:1215–1224. https://doi.org/ 10.1111/cobi.12331
- Webb NP, Pierre C (2018) Quantifying anthropogenic dust emissions. Earth's. Future 6:286–295. https:// doi.org/10.1002/2017EF000766
- WHO (World Health Organization) (2016) Ambient air pollution: a global assessment of exposure and burden of disease. https://apps.who.int/iris/bitstream/handle/10665/250141/9789241511353-eng. pdf?sequence=1&isAllowed=y. accessed in 2020
- Wiggs GFS (2011) Geomorphological hazards in drylands. In: Thomas DSG (ed) Arid zone geomorphology. John Wiley & Sons Ltd, Chichester, UK, pp 583–598. https://doi.org/10.1002/97804 70710777.ch23
- WMO (World Meteorological Organization) (2021) Dust Storm or Sandstorm. https://cloudatlas.wmo. int/en/dust-storm-or-sandstorm.html#:~:text=Definition%3A%20Dust%20storm%20or%20san dstorm,with%20loose%20dust%20or%20sand. accessed in 2020
- Woods-Ballard B, Wilson S, Udale-Clarke H, Illman S, Scot T, Ashley R, Kellagher R (2015) The SuDS Manual. CIRIA, London
- Wu X, Bennett DH, Lee K, Cassady DL, Ritz B, Hertz-Picciotto I (2011) Longitudinal variability of timelocation/activity patterns of population at different ages: a longitudinal study in California. Environ Health 10:80. https://doi.org/10.1186/1476-069X-10-80
- Xin-fa Q, Yan Z, Qi-long M (2001) Sand-dust storms in China: temporal-spatial distribution and tracks of source lands. J Geogr Sci 11:253–260. https://doi.org/10.1007/BF02892308
- Yarza S, Hassan L, Shtein A, Lesser D, Novack L, Katra I, Kloog I, Novack V (2020) Novel approaches to air pollution exposure and clinical outcomes assessment in environmental health studies. Atmosphere 11:122. https://doi.org/10.3390/atmos11020122
- Yayneshet T, Eik LO, Moe SR (2009) The effects of exclosures in restoring degraded semi-arid vegetation in communal grazing lands in northern Ethiopia. J Arid Environ 73:542–549. https://doi.org/10. 1016/j.jaridenv.2008.12.002
- Yoshioka M, Mahowald NM, Conley AJ, Collins WD, Fillmore DW, Zender CS, Coleman DB (2007) Impact of desert dust radiative forcing on sahel precipitation: relative importance of dust compared to sea surface temperature variations, vegetation changes, and greenhouse gas warming. J Clim 20:1445–1467. https://doi.org/10.1175/JCLI4056.1
- Young H, Fitzpatrick M, Mashak A, Radday A, Staro F, Aishwarya V (2019) Lessons for Taadoud II: improving natural resource management, https://reliefweb.int/sites/reliefweb.int/files/resources/Taado udIIDeskStudy2019-6.26FINAL.pdf. accessed 2021
- Zender CS (2003) Mineral dust entrainment and deposition (DEAD) model: description and 1990s dust climatology. J Geophys Res 108:4416. https://doi.org/10.1029/2002JD002775
- Zieba FW, Yengoh GT, Tom A (2017) Seasonal migration and settlement around lake chad: strategies for control of resources in an increasingly drying lake. Resources 6:41. https://doi.org/10.3390/resou rces6030041
- Zobeck TM, Van Pelt RS (2015) Wind Erosion. In: Hatfield JL, Sauer TJ (eds) Soil management: building a stable base for agriculture. Soil Science Society of America, Madison, WI, USA, pp 209–227. https:// doi.org/10.2136/2011.soilmanagement.c14
- Zota AR, Willis R, Jim R, Norris GA, Shine JP, Duvall RM, Schaider LA, Spengler JD (2009) Impact of mine waste on airborne respirable particulates in Northeastern Oklahoma, United States. J Air Waste Manag Assoc 59:1347–1357. https://doi.org/10.3155/1047-3289.59.11.1347
- Zucca C, Middleton N, Kang U, Liniger H (2021) Shrinking water bodies as hotspots of sand and dust storms: the role of land degradation and sustainable soil and water management. CATENA 207:105669. https://doi.org/10.1016/j.catena.2021.105669

Publisher's note Springer Nature remains neutral with regard to jurisdictional claims in published maps and institutional affiliations.